

Notice of Meeting:

I hereby give notice that an extraordinary Meeting of the Council will be held on:

Date: Tuesday 11 June 2024

Time: 11:30 am

Meeting Room: Council Chamber and Audio Visual link

Venue: Municipal Building, Garden Place, Hamilton

Lance Vervoort Chief Executive

Council Kaunihera OPEN AGENDA

Membership

Chairperson *Heamana*

Mayor Paula Southgate

Deputy Chairperson

Heamana Tuarua

Deputy Mayor Angela O'Leary

Members Cr Maxine van Oosten Cr Geoff Taylor

Cr Moko Tauariki Cr Sarah Thomson
Cr Ewan Wilson Cr Emma Pike
Cr Mark Donovan Cr Anna Casey-Cox
Cr Louise Hutt Cr Kesh Naidoo-Rauf

Cr Andrew Bydder Vacancy

Quorum: A majority of members (including vacancies)

Meeting Frequency: Monthly – or as required

Amy Viggers Mana Whakahaere Governance Lead

7 June 2024

Telephone: 07 838 6699 Amy.Viggers@hcc.govt.nz www.hamilton.govt.nz

Purpose

The Council is responsible for:

- 1. Providing leadership to, and advocacy on behalf of, the people of Hamilton.
- 2. Ensuring that all functions and powers required of a local authority under legislation, and all decisions required by legislation to be made by local authority resolution, are carried out effectively and efficiently, either by the Council or through delegation.

Terms of Reference

- 1. To exercise those powers and responsibilities which cannot legally be delegated by Council¹:
 - a) The power to make a rate.
 - b) The power to make a bylaw.
 - c) The power to borrow money, or purchase or dispose of assets, other than in accordance with the Long Term Plan.
 - d) The power to adopt a Long Term Plan or Annual Plan, or Annual Report.
 - e) The power to appoint a Chief Executive.
 - f) The power to adopt policies required to be adopted and consulted on under the Local Government Act 2002 in association with the Long Term Plan or developed for the purpose of the Council's Governance Statement.
 - g) The power to adopt a remuneration and employment policy.
 - h) The power to approve or change the District Plan, or any part of that Plan, in accordance with the Resource Management Act 1991.
 - i) The power to approve or amend the Council's Standing Orders.
 - j) The power to approve or amend the Code of Conduct for Elected Members.
 - k) The power to appoint and discharge members of committees.
 - I) The power to establish a joint committee with another local authority or other public body.
 - m) The power to make the final decision on a recommendation from the Parliamentary Ombudsman, where it is proposed that Council does not accept the recommendation.
 - n) The power to amend or replace the delegations in Council's *Delegations to Positions Policy*.
- 2. To exercise the following powers and responsibilities of Council, which the Council chooses to retain:
 - a) Resolutions required to be made by a local authority under the Local Electoral Act 2001, including the appointment of an electoral officer and reviewing representation arrangements.
 - b) Approval of any changes to Council's vision, and oversight of that vision by providing direction on strategic priorities and receiving regular reports on its overall achievement.
 - c) Approval of any changes to city boundaries under the Resource Management Act 1991.
 - d) Adoption of governance level strategies plans and policies which advance Council's vision and strategic goals.

¹ Clause 32, Schedule7, Local Government Act 2002

- e) Approval of the Triennial Agreement.
- f) Approval of the local governance statement required under the Local Government Act 2002.
- g) Approval of a proposal to the Remuneration Authority for the remuneration of Elected Members.
- h) Approval of any changes to the nature and delegations of the Committees.
- i) Approval or otherwise of any proposal to establish, wind-up or dispose of any holding in, a CCO, CCTO or CO.
- j) Approval of city boundary changes, including in respect of Strategic Boundary Land Use Agreements.
- k) Approval of Activity Management Plans.
- I) Sister City relationships.

Oversight of Strategies, Plans and Reports:

- Long Term Plan
- Annual Plan
- Annual Report
- Shaping Hamilton Kirikiriroa Together
- Our Climate Future
- He Pou Manawa Ora

Oversight of Policies and Bylaws:

- Corporate Hospitality and Entertainment Policy
- Delegations to officers specific to the Resource Management Act 1991
- Delegations to Positions Policy
- Elected Members Support Policy
- Significance and Engagement Policy
- Climate Change Policy
- Any Community Engagement Policies

ITEM	TABLE OF CONTENTS	PAGE
1	Apologies – Tono aroha	5
2	Confirmation of Agenda – Whakatau raarangi take	5
3	Declarations of Interest – Tauaakii whaipaanga	5
4	Public Forum – Aatea koorero	5
5	Chair's Report (Verbal)	6
6	Submission on the Local Government Water Services Preliminary Arrangements Bill	7

1 Apologies – Tono aroha

2 Confirmation of Agenda – Whakatau raarangi take

The Council to confirm the agenda.

3 Declaration of Interest – Tauaakii whaipaanga

Members are reminded of the need to be vigilant to stand aside from decision making when a conflict arises between their role as an elected representative and any private or other external interest they might have.

4 Public Forum – Aatea koorero

As per Hamilton City Council's Standing Orders, a period of up to 30 minutes has been set aside for a public forum. Each speaker during the public forum section of this meeting may speak for five minutes or longer at the discretion of the Mayor.

Please note that the public forum is to be confined to those items falling within the terms of the reference of this meeting.

Speakers will be put on a Public Forum speaking list on a first come first served basis in the Council Chamber prior to the start of the Meeting. A member of the Governance Team will be available to co-ordinate this. As many speakers as possible will be heard within the allocated time.

If you have any questions regarding Public Forum please contact Governance by telephoning 07 838 6699.

Item 5

Council Report

Committee: Council **Date:** 11 June 2024

Author: Amy Viggers **Authoriser:** Amy Viggers

Position: Governance Lead **Position:** Governance Lead

Report Name: Chair's Report (Verbal)

Report Status	Open
---------------	------

1. Mayor Southgate will present a verbal report to the Council concerning Dunedin City Council request for support for a remit regarding the rates rebate to be taken to LGNZ's Annual General Meeting in August 2024.

Recommendation – Tuutohu

- 2. That the Council:
 - a) receives the report; and
 - b) supports/does not support Dunedin City Council's remit regarding the rates rebate.

Attachments - Ngaa taapirihanga

There are no attachments for this report.

Council Report

Committee: Council **Date:** 11 June 2024

Author: Andrew Parsons **Authoriser:** Andrew Parsons

Position: General Manager **Position:** General Manager

Infrastructure and Assets Infrastructure and Assets

Report Name: Submission on the Local Government Water Services Preliminary

Arrangements Bill

Report Status	Open
---------------	------

Purpose - Take

1. To inform the Council of the Local Government (Water Services Preliminary Arrangements) Bill.

2. To seek approval from the Council of the submission on to the Finance and Expenditure Select Committee on the Local Government (Water Services Preliminary Arrangements) Bill in time for the submission deadline of 13 June 2024.

Staff Recommendation - Tuutohu-aa-kaimahi

- 3. That the Council:
 - a) receives the report;
 - b) delegates to the Mayor and Deputy Mayor authority to approve the submission to the to the Finance and Expenditure Select Committee on the Local Government (Water Services Preliminary Arrangements) Bill reflecting any feedback from this meeting.

Executive Summary - Whakaraapopototanga matua

- 4. The Local Government (Water Services Preliminary Arrangements) Bill (the Bill) was introduced to Parliament on 4 June 2024. The Bill has been referred to the Finance and Expenditure Select Committee. Submissions on the Bill close on 13 June 2024. There will be a truncated Select Committee Process.
- 5. The Bill is the second step in the Government's process to implement its Local Water Done Well programme and marks a significant shift from the approach of the previous Government.
- 6. The Bill sets out the requirements for a Water Services Delivery Plan and the initial step towards economic regulation through the Commerce Commission through an information disclosure regime. There are a number of detailed matters which warrant attention and consideration through the Select Committee process. Those matters are addressed in the draft submission.

- 7. More importantly, the Bill does not address many of the fundamental characteristics of a water services council-controlled organisation. Without addressing these Council is very limited in its ability to fully consider alternative water delivery models. The draft submission urges the Government to address these issues with urgency and provide the clarity needed for Council to find the best waters delivery outcomes for Hamilton.
- 8. The Minister of Local Government has also released an amendment paper relating to the way in which Taumata Arowai deals with the National Policy Statement for Freshwater Management.

Background - Koorero whaimaarama

- 9. Council has previously been briefed on the intent of the Government's Local Water Done Well programme. The Bill is the second piece of legislation intended to support local decision making over the delivery of water services and address the significant water delivery and investment challenges that Councils face.
- 10. A third Bill is expected later this year to address the critical issues of the scope, authority and powers of a water services council-controlled organisation. That Bill will also move beyond the initial disclosure requirements set out in this Bill to establish the ongoing framework for the operation and regulation of water services delivery.
- 11. A summary of the Bill is attached to this paper as **attachment 1**.
- 12. The full text of the Bill is attached to this report as attachment 2.
- 13. The full text of the Minister's amendment paper is attached to this report as attachment 3.

Discussion - Matapaki

- 14. As expected, and foreshadowed by the Minister, the Bill is a step in the process of the Government's Local Water Done Well programme. It does not address many of the fundamental issues associated with what a water services council-controlled organisation would look like, or the authority with which it could operate. It is a step toward the establishment of a permanent and ongoing framework.
- 15. The draft submission argues that to find the best outcomes for Hamilton we need to consider a range of possible future delivery options and potential partnerships with other Councils. Yet at this stage we are still waiting for critical policy decisions and legislation that will shape what a Water Services Council-Controlled Organisation (WSCCO) looks like, their powers and authorities, and the way in which Councils relate to them.
- 16. Key issues include the statutory powers and authorities of a waters CCO, the ability to access property, control connections, make use of bylaws (or not), manage trade wastes, act as a requiring authority under the Resource Management Act and Public Works Act, the charging mechanisms that will be available to a WSCCO, and the taxation status of a WSCCO. These are material matters in relation the legal ability of a waters CCO to undertake functions currently undertaken by the Council. They are also material matters in relation to the cost-effectiveness of alternative delivery models.
- 17. A change in the tax status of delivery of water services would increase costs significantly, with no benefit to ratepayers or water users.
- 18. The draft submission argues that legislative clarity is needed quickly in order to support the decisions that Councils will need to make as they prepare their Water Services Delivery Plans. Urging the government to expedite resolution of these issues and provide clarity as soon as possible.

- 19. The Bill establishes the new requirement of a Water Services Delivery Plan and the clear goal of ensuring that the delivery of waters services is financially sustainable by 30 June 2028. The draft submission recognises the importance of this step but reinforces the need for any new regulation to be cost effective and for the benefits to exceed the cost of compliance.
- 20. The Bill will enable Councils to prepare water services delivery plans either by themselves or jointly with other councils. The Bill recognises that the joint delivery of activity may not apply across all three waters. However, irrespective of whether for instance stormwater is being dealt with by councils and water supply and wastewater through a joint arrangement, the joint water services delivery plan must include all aspects of delivery across all three waters.
- 21. The draft submission recognises that there are significant costs associated with any change to delivery of waters services and seeks government financial assistance in meeting those costs.
- 22. The Bill defines in law the water supply network, the wastewater network, and the stormwater network. There are issues with the definition of the stormwater network proposed in the Bill. The piped stormwater network is relatively easy to identify. But what surrounds that is considerably more difficult.
- 23. Over years streams have been both piped as part of development, and then day-lighted as part of more modern approaches to urban development and stormwater management. It is not at all clear from the definition in the Bill which parts constitute the stormwater network. Hamilton City Council's parks and reserves are key parts of our communities' open space and recreation facilities, they also perform critical stormwater management functions.
- 24. Roads are critical both in terms of their impact on generating stormwater runoff and it the connections that they provide to both the piped and un-piped flow of stormwater. It is not at all clear from the definition in the Bill whether they are part of the stormwater network.
- 25. Equally, most modern urban developments include significant stormwater detention facilities, wetlands, tanks, and dams. Some of these become Council property when the development is complete, others remain in private ownership. Functionally these are critical to the effective management of stormwater and flooding risks, but these are excluded from the definition of stormwater proposed in the Bill. The draft submission urges the Select Committee to consider a more careful definition of the stormwater network.
- 26. For Hamilton, one of the key factors that is impacting on the ability of the Council to fund waters services is the need to build very substantial new infrastructure to support growth. The Bill provides for a link to supporting housing growth and urban development, but connects that to the Long-Term Plan, whereas the Future Development Strategy under the National Policy Statement for Urban Development is a more comprehensive statement of the way in which the Council is responding to growth and development pressures. The draft submission seeks to address this.
- 27. The Bill provides a mechanism for the Minister to appoint a Crown facilitator or a Crown water services specialist in some circumstances. Critically, the circumstances for appointment do not include an appointment to a Council that has decided to proceed by itself and has a water services delivery plan that meets the statutory requirements. This is important because it means that a Minister cannot appoint a Crown facilitator or a Crown water services specialist to one compliant council as a way of addressing issues or failing in another non-compliant council. It is for this reason that the draft submission generally supports the provisions in the Bill relating to these appointments.
- 28. The Bill includes a number of provisions that are specific to Watercare. The draft submission does not address these other than to note concern that these do not become the default national solution because they may result in a poor outcome for Hamilton City Council. For instance, there is a considerable risk that the Watercare approach to the potential liability for

- debts would result in a more difficult and more costly borrowing environment for any other (smaller) new waters entity.
- 29. The Watercare provisions also include an ongoing role for a Crown Monitor which would be expensive in the Hamilton context. The draft submission emphasises the need to ensure that any changes to delivery arrangements deliver better outcomes not more costly solutions for our ratepayers. It notes our desire to work with the Government to ensure that the next step in water reform does indeed deliver benefits for our communities.
- 30. The Bill provides for some alternative and more streamlined consultation and decision-making processes for establishing water services council-controlled organisations. Whilst this is a helpful step, it is unlikely that the new provisions would meet the expectations of either the Council or the Hamilton community in relation to decisions that go to the heart of the ownership, funding, and delivery of waters services.
- 31. The Bill enables Council to continue to use the normal provisions of the Local Government Act 2002 if they so choose. It is also possible that the new provisions will not completely remove the requirement to consult (for instance on a major change to the Long-Term Plan) and therefore may not by themselves provide the most transparent way of engaging with the Hamilton community. The draft submission generally supports these and other provisions, recognising that the Bill retains significant local choices in the way in which Council chooses to engage with its community.
- 32. The Minister's amendment paper sets out minor changes to the Waters Services Act 2021 in relation to the powers of Taumata Arowai to establish environmental performance standards relating to wastewater. The proposed amendment removes from consideration the hierarchy of obligations relating to Te Mana o te Wai set out in the National Policy Statement for Freshwater Management (NPSFM) when Taumata Arowai is making wastewater environmental performance standards.
- 33. The amendment further provides for the future repeal of this new section when a change is made to the NPSFM. This matter is not addressed in the draft submission because it is considered to be a short-term technical change whilst the Government considers a broader change to the NPSFM. The change only relates to considerations in relation to wastewater environmental performance standards. The substance of the issues relating to the way in which Te Mana o te Wai is reflected in regulation can be addressed though consultation on changes to the NPSFM. We note that the impact and standing of Te Ture Whaimana as a National Policy Statement is unchanged by this proposed amendment.
- 34. The draft submission is attached to this report **Attachment 4**.

Options

- 35. The simple choice that the Council has currently is whether or not to make submissions on the Bill.
- 36. Given the importance of water services delivery to Hamilton and to Hamilton City Council, it is strongly recommended that the Council take this opportunity to address the issues that have been identified in the Bill, and to also impress upon the Government the need for it to address the significant policy and legislative issues that must still be resolved in order for the Council to make sound long-term decisions over the future delivery of water services to Hamilton.

Financial Considerations - Whaiwhakaaro Puutea

37. The Bill requires the Council to prepare a water services delivery plan. The full scope of such a plan will not be known until any rules established by the Secretary for Local Government are made. This means that the full costs of preparing a plan are not yet known.

- 38. One of the key themes of the draft submission is to ensure that any new regulations or disclosure requirements are cost effective and that the benefits of the regulation exceed the cost of compliance.
- 39. Council made some financial provision for preparing a plan at the <u>30 May 2024 meeting</u> when it resolved to reallocate some Better Off and Transition Funding.

Legal and Policy Considerations - Whaiwhakaaro-aa-ture

40. Staff confirm that recommendations comply with the Council's legal and policy requirements.

Climate Change Impact Statement

- 41. Staff have assessed these matters addressed in this report and determined that no adaptation assessment is required.
- 42. Staff have assessed these matters addressed in this report and determined that no change in greenhouse emissions will occur as a result of this report.

Wellbeing Considerations - Whaiwhakaaro-aa-oranga tonutanga

- 43. The purpose of Local Government changed on the 14 May 2019 to include promotion of the social, economic, environmental and cultural wellbeing of communities in the present and for the future ('the 4 wellbeings').
- 44. The subject matter of this report has been evaluated in terms of the 4 wellbeings during the process of developing this report.
- 45. The recommendations set out in this report are consistent with that purpose.

Risks - Tuuraru

- 46. The Bill introduces new compliance related risks to Council in relation to the provision of waters services. The most significant risk arises through the creation of new powers of Ministerial intervention, that include the ability for a Crown Water Services Specialist to direct the Council in relation to a water services delivery plan. Council has at its disposal ample means of mitigating this risk through compliance with the legislative requirements of the Bill. Provided the Council complies there would be no grounds for a Ministerial intervention.
- 47. The other areas of risk to Council relate to the timing of any rules made by the Secretary for Local Government, the nature of any disclosure requirement made by the Commerce Commission, these matters are addressed in the draft submission.
- 48. The most significant risk arising from the Bill is extent to which the disclosures highlight any inability by the Council to achieve financial sustainability for waters by 30 June 2028. This is not a new risk, in that finding financial sustainable delivery for waters is one of the key issues that the Council is grappling with.

Significance & Engagement Policy - Kaupapa here whakahira/anganui

- 49. Staff have considered the key considerations under the Significance and Engagement Policy and have assessed that the matter(s) in this report has/have a low level of significance.
- 50. Council's submission position takes into account community views that have been expressed through Council's previous engagements on water reform.
- 51. Individuals will have the opportunity to make submissions to the Select Committee by 13 June 2024.

52. There is no time for further community engagement given the timetable adopted by the Select Committee for submissions.

tem (

53. Given the low level of significance determined, the engagement level is low. No engagement is required.

Attachments - Ngaa taapirihanga

- Attachment 1 A Summary of the Local Government (Water Services Preliminary Arrangements) Bill
- Attachment 2 The full text of the Local Government (Water Services Preliminary Arrangements) Bill
- Attachment 3 The full text of the Local Government (Water Services Preliminary Arrangements) Bill Amendment Paper
- Attachment 4 Hamilton City Council Draft Submission Local Government Water Services .

Local Government (Water Services Preliminary Arrangements) Bill

Initial analysis summary

Council Agenda 11 June 2024- OPEN Page 13 of 109

Context to Bill

Policy objective: lay foundation for a new framework of water services management and financially sustainable delivery models that meet regulatory standards, including a new water services delivery arrangement for Auckland.

What does this Bill provide for:

- Water services delivery plans to be submitted within 12 months of the Bill's enactment. Extensions will be
 available based on stipulated criteria, including extra time to consult on joint arrangements or finalise
 negotiations, and subject to any reasonable conditions.
- The information on the current state of water services will lay the foundation for information disclosure as part of a future comprehensive economic regulation regime.
- Councils can combine to prepare a joint plan to reflect joint arrangements and how water supply, wastewater, and stormwater services will be provided throughout the relevant districts. There is flexibility whether to include some or all stormwater services in a joint arrangement to enable councils to make the right decisions for their local circumstances.

Next steps: A further Bill, to be introduced in December 2024, will set out a comprehensive range of options, tools and models that will enable councils to exercise those choices, contingent on meeting criteria for financial sustainability.

Council Agenda 11 June 2024- OPEN Page 14 of 109

Water services plan overview

On a high level, the Plan must:

- 1. provide detailed information on the current state of councils' water services
- 2. demonstrate publicly the commitment to deliver water services in a way that
 - (i) ensures that the council will meet all relevant regulatory quality standards for its stormwater network, wastewater network, and water supply network; and
 - (ii) is financially sustainable for the council; and
 - (iii) supports the council's housing growth and urban development, as specified in its long-term plan.

Plan must relate to all water services in the relevant area (whether a single council or joint water plan across districts)

Council Agenda 11 June 2024- OPEN Page 15 of 109

Water services plan detail - 14 categories

1. Current state of network described	8. Asset management approach described
2. Current levels of service described	Issues, constraints and risks to be described, together with impact on delivery
3. Areas that receive water services in district and those that don't described, together with infrastructure associated with provision for population growth and development capacity	10. Anticipated or proposed model for delivering water services (including any joint arrangement or to continue to deliver alone)
4. Whether water services comply with regulatory requirements (and to what extent)	11. How will revenue be separated from other functions
5. Details of capex and opex required to deliver water services and for regulatory requirements	12. What consultation undertaken to developed information included at 10 and 11 above (model and revenue)
 Financial projections for period of plan (refer next slide) – opex, revenue, capex, debt. 	13. Plan to make financially sustainable by 30 June 2028
7. Assessment of current condition, lifespan and value of network	14. Any other information set in rules by Secretary for Local Government under this Act

Council Agenda 11 June 2024- OPEN Page 16 of 109

Further information for a joint plan

- 1. Who are you joining with?
- 2. What is the scope of the joint arrangement? Is it 2 waters or 3 waters or 2 waters for some and three waters for others.
- 3. What form of arrangement e.g. a joint water services CCO or other arrangement under section 137 LGA. If available at time of submission of plan:
 - > indicative implementation plan and timeline
 - > further information about the joint arrangement, including the ownership structure, the governance structure, and the control and financial rights of each territorial authority in the joint arrangement.

Council Agenda 11 June 2024- OPEN Page 17 of 109

Further information Secretary can request

Requirements for Plan can be extended by Secretary for Local Government to include information that:

- 1. improves accountability to council's community
- 2. provides a basis for regulating the delivery of water services
- 3. relates to 1 or more of the following:
 - · financial matters (including, for example, revenues, equity levels, debt arrangements, and expenses)
 - the assets involved in delivering water services (including, for example, asset management plans and asset replacement policies)
 - · financial and non-financial performance measures
 - · the relevant performance measures and statistics relating to water quality
 - pricing practices, assumptions, policies and methodologies used in delivering water services.

Before making a rule, the Secretary must consult each person or organisation that the Secretary considers to hold views that are representative, or that may be held, in the local government sector.

Council Agenda 11 June 2024- OPEN Page 18 of 109

Period covered by plan

- 1. Not less than 10 consecutive financial years, starting with the 2024-25 financial year.
- 2. Requires detailed information to be provided for each of the first 3 financial years
- 3. An outline in relation to each of the subsequent financial years covered by the plan

Council Agenda 11 June 2024- OPEN Page 19 of 109

Process for adopting Plan and Minister's powers

- A council must adopt a water services delivery plan by resolution.
- Is to be submitted to the Secretary for Local Government for acceptance
- Certification by CE(s) that (a) the plan complies with the Act; and (b) the information contained in the plan is true and accurate.
- If Secretary not satisfied with plan can decide not to accept or ask for it to be amended and resubmitted by a specified date.
- Minister can appoint Crown Facilitator if asked (and Minister agrees) or Minister believes it would be beneficial because plan unlikely to be submitted or a group is having difficult agreeing on the terms of a joint plan. Can also appoint Crown water services specialist.
- Crown Facilitator can assist with preparation, advise on preparation direct on how to prepare or amend (where Secretary requests it amended under Act). Assist councils in reaching agreement on joint plan (where requested). Can recommend to Minister appointment of Crown water services specialist of further action under Part 10 LGA.
- Crown water services specialist can direct a council or group to appoint a specified plan

Council Agenda 11 June 2024- OPEN Page 20 of 109

What does financially sustainable mean?

Financially sustainable means, in relation to council's delivery of water services, that:

- (a) the revenue applied to the authority's delivery of those water services is sufficient to ensure the authority's long-term investment in delivering water services; and
- (b) the authority is financially able to meet all regulatory standards and requirements for the authority's delivery of those water services

Council Agenda 11 June 2024- OPEN Page 21 of 109

Alternative consultation

- Under the alternative arrangements, councils may:
 - identify 2 options for delivering water services (the status quo and the proposed new arrangement) in place of the requirement to identify all reasonably practicable options:
 - consult only once and are not required to consult additionally on an amendment to the long-term plan resulting from a decision relating to water services councilcontrolled organisations:
 - conditionally approve a long-term plan amendment subject to corresponding agreement from the other territorial authorities that are parties to a joint water services council-controlled organisation:
 - consider in their decision making the impact of a joint water services council controlled organisation on communities in the area covered, as well as in the
 council's district.
- A joint committee between councils wanting to work together can be set up to undertake 1 or more of the process steps.
- Councils will be temporarily exempted from the cost-effectiveness review laid out in section 17A of the Local Government Act 2002 that relates to a decision to establish, join, or amend a water services council-controlled organisation.

Council Agenda 11 June 2024- OPEN Page 22 of 109

Auckland

Auckland Council's preferred model envisages greater financial independence for Watercare by:

- shifting the legislative obligation to provide water supply and wastewater services from Auckland Council to Watercare:
- prohibiting Auckland Council from providing financial support to Watercare in any situation or under any conditions:
- enabling the Minister of Local Government to appoint a Crown review team, Crown monitor, or Crown manager to Watercare if a significant problem exists:
- introducing interim economic regulation under the oversight of a Crown monitor until further legislation establishes a long-term economic regulatory framework:
- enabling the Crown to reimburse Watercare for expenses incurred during an emergency.

The provisions enabling Auckland Council's preferred model for water services delivery will commence on a date set by Order in Council made by the Minister of Local Government or, at latest, 1 July 2025. The provisions relating to interim economic regulation will commence on the day after Royal assent.

Auckland Council's water services delivery plan will only need to cover stormwater services.

Council Agenda 11 June 2024- OPEN Page 23 of 109

Local Government (Water Services Preliminary Arrangements) Bill

Government Bill

Explanatory note

General policy statement

This Bill establishes preliminary arrangements for local government water services delivery. It is an omnibus Bill that amends other Acts and is introduced in accordance with Standing Order 267(1)(a) as the amendments deal with an interrelated topic that can be regarded as implementing a single broad policy. The single broad policy for this Bill is to lay the foundation for a new framework of water services management and financially sustainable delivery models that meet regulatory standards, including a new water services delivery arrangement for Auckland. Councils will be able to use the provisions to start planning future water services delivery and undertake steps to establish, join, or amend council-controlled organisations from the day after Royal assent

The Bill contains stand-alone provisions and also amends—

- the Local Government Act 2002; and
- the Local Government (Auckland Council) Act 2009; and
- the Local Government (Rating) Act 2002; and
- the Receiverships Act 1993; and
- the Civil Defence and Emergency Management Act 2002.

Introduction

New Zealand's long-standing water infrastructure challenges require locally led and financially sustainable solutions. The Water Services Acts Repeal Act 2024 (the **Repeal Act**) put an end to the proposed 10 water services entities and confirmed that councils will continue to own and be responsible for water services. This Bill requires councils to provide transparent information about the current state of water services and to set out proposals to achieve financially sustainable water services that meet

Attachment 2

regulatory standards. With greater oversight and support from the Government, councils will have the flexibility and discretion to determine the optimal structure, delivery method, and funding and financing arrangements for their water services.

A further Bill, to be introduced in December 2024, will set out a comprehensive range of options, tools and models that will enable councils to exercise those choices, contingent on meeting criteria for financial sustainability.

Relationship to other proposed legislation

There are 3 legislative components to achieving the new framework for water services delivery. The Repeal Act reversed the previous approach to reforming water services delivery. This Bill is the second component and sets out the preliminary arrangements to put water services infrastructure on the path to long-term financial sustainability.

Further legislation will provide for the long-term replacement regime, including—

- long-term requirements for financial sustainability:
- establishing new classes of council-controlled water organisations and service delivery models:
- accountability, planning, and reporting regimes for water services:
- providing for comprehensive economic regulation (being developed in conjunction with the Minister of Commerce and Consumer Affairs):
- amendments to the regulatory settings for Taumata Arowai–the Water Services Regulator:
- establishing a regulatory backstop power:
- refinements to water services delivery system settings:
- detailed changes to the Local Government Act 2002 and other legislation to strengthen the delivery of water services.

Water services delivery plans

The Bill requires territorial authorities to submit water services delivery plans (**plans**) within 12 months of the Bill's enactment.

The plans will provide detailed information on the current state of councils' water services arrangements for water supply (including drinking water), wastewater, and stormwater and set out a strategy for how they will achieve the delivery of financially sustainable water services and meet regulatory quality standards. The information on the current state of water services will lay the foundation for information disclosure as part of a future comprehensive economic regulation regime.

Territorial authorities may combine to prepare a joint plan to reflect joint arrangements. A joint plan must explain how water supply, wastewater, and stormwater services will be provided throughout the districts included in the joint plan. There is flexibility whether to include some or all stormwater services in a joint arrangement

Explanatory note

to enable territorial authorities to make the right decisions for their local circumstances.

The plans must be provided to the Secretary for Local Government within 12 months of the Bill's enactment for consideration as to whether they comply with the requirements set out in the Bill. Extensions will be available based on criteria in the Bill, including extra time to consult on joint arrangements or finalise negotiations, and subject to any reasonable conditions.

Additional information disclosure

Foundational information disclosure will be provided through the plans to lay the groundwork for comprehensive economic regulation. The Bill provides for additional information disclosure for territorial authorities and council-controlled organisations providing water services where such disclosure promotes the long-term benefit of consumers and supports efficiency, innovation, and investment.

The additional information disclosure requirements will be applied to territorial authorities or water services council-controlled organisations by Order in Council made on the recommendation of the Minister of Local Government and the Minister of Commerce and Consumer Affairs. Determinations made by the Commerce Commission will outline the content of the additional information disclosure.

Alternative consultation and decision-making process for water services councilcontrolled organisations

The Bill sets out optional alternative consultation and decision-making requirements for territorial authorities to use if they wish to when establishing, joining, or amending a water services council-controlled organisation. Territorial authorities have the flexibility to use any or all of the alternative requirements instead of the existing processes in the Local Government Act 2002. If they do not use the alternative requirements in this Bill, the relevant requirements in the Local Government Act 2002 will continue to apply.

Under the alternative arrangements, territorial authorities may—

- identify 2 options for delivering water services (the status quo and the proposed new arrangement) in place of the requirement to identify all reasonably practicable options:
- consult only once and are not required to consult additionally on an amendment to the long-term plan resulting from a decision relating to water services council-controlled organisations:
- conditionally approve a long-term plan amendment subject to corresponding agreement from the other territorial authorities that are parties to a joint water services council-controlled organisation:
- consider in their decision making the impact of a joint water services councilcontrolled organisation on communities in the area covered, as well as in the territorial authority's district.

A joint committee between territorial authorities wanting to work together can be set up to undertake 1 or more of the process steps.

Local Government (Water Services Preliminary Arrangements) Bill

Councils will be temporarily exempted from the cost-effectiveness review laid out in section 17A of the Local Government Act 2002 that relates to a decision to establish, join, or amend a water services council-controlled organisation.

New financially sustainable model for Watercare

The Bill enables Auckland Council to implement its preferred model for water services delivery. Watercare Services Limited (**Watercare**) is a registered company and 100% owned by Auckland Council. It provides water supply and wastewater services in Auckland. Auckland Council's preferred model envisages greater financial independence for Watercare by—

- shifting the legislative obligation to provide water supply and wastewater services from Auckland Council to Watercare:
- prohibiting Auckland Council from providing financial support to Watercare in any situation or under any conditions:
- enabling the Minister of Local Government to appoint a Crown review team, Crown monitor, or Crown manager to Watercare if a significant problem exists:
- introducing interim economic regulation under the oversight of a Crown monitor until further legislation establishes a long-term economic regulatory framework:
- enabling the Crown to reimburse Watercare for expenses incurred during an emergency.

The provisions enabling Auckland Council's preferred model for water services delivery will commence on a date set by Order in Council made by the Minister of Local Government or, at latest, 1 July 2025. The provisions relating to interim economic regulation will commence on the day after Royal assent.

Auckland Council's water services delivery plan will only need to cover stormwater services.

Secondary legislation

The Bill contains provision for delegated decision-making powers through secondary legislation that—

- enable the Secretary for Local Government to make rules for additional matters to be included in water services delivery plans:
- subject councils to enhanced information disclosure requirements for their water services in addition to the information provided in water services delivery plans. This will be given effect to by an Order in Council:
- enable the Commerce Commission to make determinations to set out the additional information that a council must disclose:

Local Government (Water Services Preliminary Arrangements) Bill

 enable the Watercare Crown monitor to make a charter that imposes obligations on Watercare as part of interim economic regulation.

Departmental disclosure statement

The Department of Internal Affairs is required to prepare a disclosure statement to assist with the scrutiny of this Bill. The disclosure statement provides access to information about the policy development of the Bill and identifies any significant or unusual legislative features of the Bill.

A copy of the statement can be found at http://legislation.govt.nz/disclosure.aspx?type=bill&subtype=government&year=2024&no=52

Regulatory impact statement

The Department of Internal Affairs produced a regulatory impact statement on 14 March 2024 to help inform the main policy decisions taken by the Government relating to the contents of this Bill.

A copy of this regulatory impact statement can be found at-

- https://www.dia.govt.nz/Resource-material-Regulatory-Impact-Statements-Index
- https://treasury.govt.nz/publications/informationreleases/ris

Clause by clause analysis

Clause 1 is the Title clause.

Explanatory note

Clause 2 provides that the majority of the Bill comes into force on the day after Royal assent. However, specified clauses come into force on a date set by Order in Council or 1 July 2025, whichever is earlier.

Part 1

Preliminary provisions

Clause 3 sets out the purpose of the Bill. The purpose is to require territorial authorities to provide information relating to their delivery of water services, to provide consultation and decision-making processes for territorial authorities to establish, join, or amend council-controlled organisations that deliver water services, and to provide for Watercare Services Limited to provide water services in Auckland in a financially sustainable manner.

Clause 4 sets out an overview of the Bill as follows:

- Part 1 includes the preliminary provisions:
- Part 2 provides for territorial authorities to prepare and submit water services delivery plans, sets out ministerial powers in relation to those plans, and requires territorial authorities to provide additional foundational information:

5

 Part 3 sets out alternative and additional consultation and decision-making requirements for territorial authorities when they are establishing, joining, or amending a water services council-controlled organisation:

Local Government (Water Services Preliminary

Arrangements) Bill

 Part 4 provides for the appointment of a Crown monitor to make a Watercare charter, to ensure the separation of Watercare Services Limited and Auckland Council. The Crown monitor will monitor and report on Watercare Services Limited.

Clause 5 defines terms used in the Bill.

Clause 6 provides for transitional savings, and related provisions, as set out in Schedule 1.

Clause 7 provides that the Bill, when enacted, will bind the Crown.

Part 2

Water services delivery plans and foundational information disclosure requirements

Subpart 1—Water services delivery plans

Water services delivery plan: obligation and contents

Clause 8 requires each territorial authority to prepare a water services delivery plan, setting out the current state of the authority's water services, and demonstrating a commitment to delivering water services in the future.

Clause 9 provides that, in certain circumstances, 2 or more territorial authorities may combine to deliver water services across the authorities' combined area.

Clause 10 provides that a water services delivery plan must relate to all water services

Clause 11 lists the information that a water services delivery plan must contain.

Clause 12 lists additional information that must be contained in a joint water services delivery plan.

Clause 13 requires a water services delivery plan to cover a period of not less than 10 consecutive financial years, starting with the 2024–25 financial year. A plan must provide more detail for the first 3 years covered by the plan than for the following 7 years.

Clause 14 provides that the Secretary for Local Government (the **Secretary**) may make rules relating to water services delivery plans.

Clause 15 requires a territorial authority to adopt its water services delivery plan by resolution.

Submission of water services delivery plan

Clause 16 requires a water services delivery plan to be submitted to the Secretary within 1 year of the Bill being enacted or by a later date specified by the Minister under clause 17.

Clause 17 provides that the Minister may, on application from a territorial authority, grant an extension to the deadline for submitting a water services delivery plan.

Acceptance of water services delivery plan

Clause 18 requires the Secretary to consider each water services delivery plan that is submitted, and to accept it if satisfied that the plan complies with the requirements set out in this Bill.

Clause 19 provides that after the Secretary accepts a water services delivery plan, the applicant must publish the plan, and the Secretary must provide a copy to the Commerce Commission and Taumata Arowai.

Subpart 2—Ministerial powers in relation to water services delivery plans

Crown facilitator for water services delivery plans

Clause 20 authorises the Minister to appoint a Crown facilitator for water services delivery plans, and sets out the grounds on which the Minister may make such an appointment.

Clause 21 requires the Minister to give notice of appointing a Crown facilitator, and sets out what information must be included in the notice.

Clause 22 sets out the role of a Crown facilitator for water services delivery plans.

Crown water services specialist

Clause 23 authorises the Minister to appoint a Crown water services specialist, and sets out the grounds on which the Minister may make such an appointment.

Clause 24 requires the Minister to give notice of appointing a Crown water services specialist, and sets out what information must be included in the notice.

Clause 25 sets out the role of a Crown water services specialist.

Clause 26 provides that any direction that a Crown water services specialist has given ceases to have effect when their appointment expires.

General provisions

Clause 27 requires a territorial authority or group of territorial authorities to co-operate with a Crown facilitator for water services delivery plans or a Crown water services specialist (a **ministerial appointee**).

Arrangements) Bill Ex

Clause 28 provides that the Minister's powers under this subpart are in addition to, not in place of, the Minister's powers under Part 10 of the Local Government Act 2002 (the **LGA2002**).

Local Government (Water Services Preliminary

Clauses 29 and 30 relate to a ministerial appointee's entitlement to remuneration and expenses.

Clause 31 specifies that a ministerial appointee is protected from liability for their acts or omissions done in good faith.

Subpart 3—Foundational information: additional disclosure requirements

Clause 32 sets out the purpose of subpart 3.

Clause 33 provides that subpart 3 applies to an entity (a **specified entity**) that is named in an Order in Council made by the Governor-General on the recommendation of the Minister and the Minister of Commerce and Consumer Affairs. Before making a recommendation, those Ministers must consider advice from the Secretary and the Commerce Commission.

Clause 34 defines the term specified entity.

Determinations

Clause 35 authorises the Commerce Commission to make a determination setting out the information that a specified entity must make publicly available and disclose to the Commerce Commission.

Clause 36 provides that a specified entity that is covered by a determination made under *clause 35* must make the required information publicly available and provide a copy of the information to the Commission.

Clause 37 lists what a determination made under clause 35 must include, what the Commission may have regard to when making a determination, and what types of information a specified entity may be required to provide.

Clause 38 provides that a determination may, in certain circumstances, require a specified entity to provide additional information that includes information in relation to goods and services that are not incidental to, or related to, delivering water services.

Clause 39 authorises a specified entity to charge the public to provide information that a determination requires the entity to provide.

Clause 40 provides that the Commerce Commission has additional monitoring and investigation powers to enable it to carry out its functions and exercise its powers under *Part 2* of the Bill.

Sharing of information

Clause 41 authorises the Commerce Commission and the department to share information with each other for the purposes of the Bill. It also limits the purposes for which such shared information may be used.

Local Government (Water Services Preliminary Arrangements) Bill

Explanatory note

Pecuniary penalty orders

Clause 42 authorises the High Court to order a person to pay a pecuniary penalty in the listed circumstances related to disclosing information under this subpart.

Clause 43 sets the maximum pecuniary penalty at \$500,000 for an individual or \$5 million in any other case. It also lists matters that the court must have regard to, when determining the amount of a pecuniary penalty.

Orders about information disclosure

Clause 44 provides that the High Court may order a specified entity to comply with an obligation relating to disclosing information under *subpart 3 of Part 2*.

Offence

Clause 45 provides that it is an offence to intentionally contravene a requirement relating to an obligation to disclose information, or to fail to comply with an order under clause 44. The maximum fine for the offence is \$200,000 for an individual, or \$1,000,000 in all other cases.

Miscellaneous provisions

Clause 46 lists certain provisions in the Commerce Act 1986 that apply for the purposes of subpart 3 of Part 3.

Amendment to Local Government Act 2002

Clause 47 provides that clause 48 amends the LGA2002.

Clause 48 replaces section 255(2) of the LGA2002 so it provides that the Minister may also exercise powers under Part 10 of the LGA2002 in relation to territorial authorities or joint arrangements under the Bill.

Part 3

Establishing water services council-controlled organisations

Preliminary provisions

Clause 49 sets out the purpose of Part 3 of the Bill, which is to provide alternative provisions and greater flexibility in relation to a territorial authority using a council-controlled organisation (a CCO) under the LGA2002 to deliver water services.

Alternative requirements

Clause 50 specifies that territorial authorities may rely on any of the alternative requirements set out in *clauses 51 to 54* in place of the corresponding obligations that would otherwise apply in the LGA2002. However, all other obligations in that Act continue to apply.

Clause 51 provides that a territorial authority is only required to identify and assess the listed 2 options, rather than all reasonably practicable options as required under section 77(1)(a) and (b) of the LGA2002.

Local Government (Water Services Preliminary

Arrangements) Bill

Clause 52 provides that a territorial authority is only required to consult once before deciding whether to establish or join a CCO. However, an authority may decide to consult further, in which case it must have regard to the listed matters. This clause applies in place of the obligation to consult set out in section 56(1) of the LGA2002.

Clause 53 provides that if a territorial authority is required to amend its long-term plan for the purposes of establishing, joining, or amending a CCO, it is not required to consult in certain circumstances. This clause applies in place of the obligation to consult in 93(5) of the LGA2002.

Clause 54 sets out the information that a territorial authority must make publicly available when consulting in relation to establishing, joining, or amending a CCO. This clause applies in place of the information requirements set out in section 82A(2) of the LGA2002.

Additional powers and exemption

Clause 55 provides that if a territorial authority is deciding whether to establish, join, or amend a CCO, it may consider additional listed factors. This clause applies despite sections 12(4) and 14(1)(g) of the LGA2002.

Clause 56 provides that 2 or more territorial authorities may use joint committees to perform 1 or more of the listed tasks.

Clause 57 authorises a territorial authority to conditionally approve amending its long-term plan if required for the purpose of establishing, joining, or amending a joint CCO with another territorial authority.

Clause 58 provides a temporary exemption from the obligation in section 17A of the LGA2002 to do a cost-effectiveness review in relation to a CCO in certain circumstances.

Clause 59 repeals clauses 58 and 59 5 years after the Bill is enacted.

Part 4 Watercare Services Limited

Crown monitor

Clause 60 provides that the Minister may appoint a Crown monitor to Watercare. A Crown monitor must be a government department, a Crown entity, or a company named in Schedule 4A of the Public Finance Act 1989.

Clause 61 requires the Minister to appoint the Crown monitor by providing notice to Watercare and to the Crown monitor.

Clause 62 sets out the role of the Crown monitor, which includes making a charter for Watercare and monitoring and reporting on Watercare's performance.

Explanatory note

11

Watercare charter

Clause 63 provides that the Crown monitor must prepare and make a Watercare charter. This clause also specifies that a charter must be made in 2 separate parts and lists what must be included in each part. A charter is secondary legislation.

Part 1 of Watercare charter

Clause 64 sets out the details of what must be contained in Part 1 of a Watercare charter: minimum service quality standards, financial performance objectives, and a customer compensation scheme.

Clause 65 requires Watercare to submit a draft business plan to the Crown monitor, and sets out the details of what must be contained in that plan. Watercare must submit the business plan within 4 months after the Crown monitor makes Part 1 of the Watercare charter.

Clause 66 requires that each business plan that Watercare submits to the Crown monitor must cover a period of at least 10 consecutive financial years, with more detail provided in respect of the first 3 years than for the rest of the period.

Clause 67 provides that when it receives a business plan from Watercare, the Crown monitor must review the plan and provide comments to Watercare, and may require Watercare to provide additional information. Watercare must then submit a finalised business plan that addresses the comments received and includes any requested additional information.

Part 2 of Watercare charter

Clause 68 sets out the details of a price-quality path that must be contained in Part 2 of a Watercare charter.

Effect of Watercare charter

Clause 69 provides that when the Crown monitor makes Part 2 of the charter, the charter is binding on Watercare. If there are any inconsistencies between the charter and Auckland Council's long-term plan, the charter prevails. When the charter applies, an agreement for services between Watercare and a customer must include information relating to the customer compensation scheme required by the charter.

Information disclosure

Clause 70 provides that the Crown monitor may require Watercare to provide any information the Crown monitor requires to perform its role. The Crown monitor may also require Watercare to certify that any information provided is true and accurate.

Crown monitor to monitor and report on performance

Clause 71 requires the Crown monitor to monitor Watercare's performance under the charter.

Clause 72 requires the Crown monitor to report annually on Watercare's performance against specified components of the charter. The Crown monitor must provide the report to Auckland Council, the Minister, and the Minister of Commerce and Consumer Affairs.

Local Government (Water Services Preliminary

Arrangements) Bill

Clause 73 requires the Crown monitor to report quarterly on its own performance to the Minister.

Crown monitor's expenses

Clause 74 provides that Watercare must reimburse the Crown for the Crown monitor's expenses incurred in monitoring Watercare.

Commerce Commission's functions and powers

Clause 75 provides the Commerce Commission with its functions and powers for the purposes of this Bill.

Enforcement

Clause 76 provides that the High Court may, if satisfied that Watercare contravened the charter (or attempted to contravene the charter), make various orders including an injunction or an order to pay a pecuniary penalty.

Clause 77 provides that the maximum pecuniary penalty payable under clause 76 is \$10,000,000 in respect of each act or omission. It also lists matters that the court must consider when determining the amount of a pecuniary penalty.

Clause 78 provides that the High Court may, if satisfied that Watercare has failed to provide required information or submit a business plan under *clause 65*, make various orders including an order to pay a pecuniary penalty.

Clause 79 provides that the maximum pecuniary penalty payable under clause 78 is \$300,000 in respect of each act or omission.

Clause 80 provides that the High Court may, in addition to a pecuniary penalty, impose a further penalty for each day of a continuing breach. The further penalty is \$500,000 each day for a breach under clause 76, or \$50,000 each day for a breach under clause 78.

Clause 81 provides for a right of appeal to the Court of Appeal in relation to proceedings under clause 76, on a question of law.

Related amendments to LGA2002

Clause 82 provides that clauses 83 to 91 amend the LGA2002.

Clauses 83 and 84 amend sections 121 and 122, to provide that the Crown is not liable for, or guarantee, any debts or liabilities of Watercare.

Clause 85 amends section 124 by inserting definitions of Auckland and Watercare for the purposes of Part 7 of the LGA2002.

Explanatory note

Local Government (Water Services Preliminary Arrangements) Bill

13

Clause 86 amends section 127, which provides a duty to ensure that communities have access to drinking water in certain circumstances. The amendment extends the duty to Watercare.

Clause 87 amends section 130, which places an obligation to provide water services on local government organisations. The amendment provides that the obligation only applies to Auckland Council in relation to stormwater services.

Clause 88 inserts new section 130A, which extends the obligation in section 130 to Watercare in relation to water supply and wastewater services.

Clauses 89, 90, and 91 amend sections 253, 254, and 255 to provide that the Ministers power's under Part 10 of the LGA2002 in relation to local authorities also extend to Watercare.

Related amendments to Local Government (Auckland Council) Act 2009

Clause 92 provides that clauses 93 to 99 amend the Local Government (Auckland Council) Act 2009.

Clause 93 amends section 4 by amending the definition of Auckland water organisation so that it means Watercare.

Clause 94 inserts new section 56A and cross-headings. New section 56A provides for the financial separation between Auckland Council and Watercare.

Clause 95 inserts new sections 57A and 57B. New section 57A requires Watercare to repay any debt to Auckland Council within a period of 5 years. New section 57B provides that new sections 57A and 57B are repealed after 5 years.

Clause 96 amends section 58 to reflect the financial separation between Auckland Council and Watercare.

Clause 97 inserts new section 60A, which relates to the situation if a receiver is appointed in respect of a loan granted to Watercare.

Clause 98 amends section 92 to reflect the financial separation between Auckland Council and Watercare.

Consequential amendments

Clause 99 and Schedule 2 set out the consequential amendments the Bill makes to primary and secondary legislation in relation to Watercare.

Hon Simeon Brown

Local Government (Water Services Preliminary Arrangements) Bill

Government Bill

Contents

		Page
1	Title	5
2	Commencement	6
	Part 1	
	Preliminary provisions	
3	Purpose	6
4	Overview	6
5	Interpretation	7
6	Transitional, savings, and related provisions	9
7	Act binds the Crown	9
	Part 2	
	Water services delivery plans and foundational information	
	disclosure requirements	
	Subpart 1—Water services delivery plans	
	Water services delivery plan: preparation and contents	
8	Territorial authority must prepare water services delivery plan	10
9	Territorial authorities may enter into arrangement to submit joint plan	10
10	Water services to be covered by water services delivery plan	11
11	Contents of water services delivery plan	11
12	Additional requirements for joint water service delivery plans	12
13	Period covered by water services delivery plan	13
14	Secretary may make rules in relation to water services delivery plans	14

52—1

15	Process to prepare and adopt water services delivery plan	14
	Submission of water services delivery plan	
16	Plan must be submitted to Secretary for acceptance	15
17	Minister may grant extension to deadline for submitting water services delivery plan	15
	Acceptance of water services delivery plan	
18	Secretary accepts water services delivery plan	16
19	Publication of accepted water services delivery plan	17
	Subpart 2—Ministerial powers in relation to water services delivery plans	
	Crown facilitator for water services delivery plans	
20	Minister may appoint Crown facilitator	17
21	How Crown facilitator appointed	18
22	Role of Crown facilitator	18
	Crown water services specialist	
23	Minister may appoint Crown water services specialist	19
24	How Crown water services specialist is appointed	20
25	Role of Crown water services specialist	21
26	Decisions and directions of Crown water services specialist	21
	General provisions	
27	Obligation to co-operate	21
28	Minister retains powers under Part 10 of LGA2002	22
29	Remuneration and expenses of ministerial appointee	22
30	Recovery of expenses and remuneration from local authority	22
31	Protection from liability	23
	Subpart 3—Foundational information disclosure requirements	
32	Purpose of this subpart	23
33	Application of this subpart	23
34	Meaning of specified entity	24
	Determinations	
35	Commission may make determination	24
36	Effect of determination	25
37	Contents of determination	25
38	Determination may require specified entity to provide additional	27
	information	
39	Charge for providing copies to public	28
40	Additional monitoring and investigation powers based on subpart 8 of Part 4 of Commerce Act 1986	28

	Sharing of information	
41	Sharing of information and documents between Commission and department	28
	Pecuniary penalty orders	
42 43	When High Court may make pecuniary penalty order Maximum amount of pecuniary penalty	29 30
	Orders about information disclosure	
44	Order requiring information disclosure requirement to be complied with	30
	Offence	
45	Offence relating to requirement to disclose information	30
	Miscellaneous provisions	
46	Application of Part 7 of Commerce Act 1986 (Miscellaneous provisions)	31
	Amendment to Local Government Act 2002	
47	Amendment to Local Government Act 2002	31
48	Section 255 amended (Application of this Part)	31
	Part 3	
	Establishing water services council-controlled organisations	
	Preliminary provisions	
49	Purpose of this Part	32
	Alternative requirements	
50	Alternatives to requirements in Local Government Act 2002	32
51	Alternative requirement: decision making	32
52	Alternative requirement: consultation	33
53	Alternative requirement: consultation on amendment to long-term plan	34
54	Alternative requirement: information requirements for consultation	34
	Additional powers and exemption	
55	Ability to consider joint service area	35
56	Joint committees	35
57	Ability to conditionally approve amending long-term plan	36
58	Exemption from cost-effectiveness review	36
59	Repeal of section 58	36
	Part 4	
	Watercare Services Limited	
	Crown monitor	
60	Minister may appoint Crown monitor	36

61 62	How Crown monitor appointed Role of Crown monitor	36 37
	Watercare charter	
63	Watercare charter	37
	Part 1 of Watercare charter	
64	Content of Part 1 of Watercare charter	38
65	Watercare must submit business plan to Crown monitor	39
66	Period covered by business plan submitted to Crown monitor	39
67	Process for finalising business plan	40
	Part 2 of Watercare charter	
68	Content of Part 2 of Watercare charter	40
	Effect of Watercare charter	
69	Effect of charter	41
	Information disclosure	
70	Crown monitor may require information disclosure	42
	Crown monitor to monitor and report on performance	
71	Crown monitor must monitor Watercare's performance	43
72	Crown monitor must make annual report	43
73	Crown monitor must make quarterly reports to Minister	43
	Crown monitor's expenses	
74	Crown monitor's expenses are recoverable from Watercare	44
	Commerce Commission's functions and powers	
75	Commerce Commission's functions and powers	44
	Enforcement	
76	High Court may impose orders	44
77	Maximum amount of pecuniary penalty	45
78	Order to disclose information	46
79	Maximum amount of pecuniary penalty	46
80	Further penalty for continuing breach	46
81	Appeal	46
	Related amendments to LGA2002	
82	Principal Act	47
83	Section 121 amended (The Crown not liable for debts)	47
84	Section 122 amended (Disclosure document and loan documents to contain statement that the Crown does not guarantee financial	47
0.5	products or loan)	
85	Section 124 amended (Interpretation)	47
86	Section 127 amended (Duty to ensure communities have access to drinking water if existing suppliers facing significant problems)	47

	Local Government (Water Services Preliminary Arrangements) Bill	cl 1
87	Section 130 amended (Obligation to maintain water services)	47
88	New section 130A inserted (Obligation to maintain water serv Watercare)	rices: 48
	130A Obligation to maintain water services: Watercare	48
89	Section 253 amended (Outline of Part)	48
90	Section 254 amended (How this Part works)	48
91	Section 255 amended (Application of this Part)	48
	Related amendments to Local Government (Auckland Council 2009	l) Act
92	Principal Act	49
93	Section 4 amended (Interpretation)	49
94	New section 56A and cross-headings inserted	49
	Auckland Council	
	56A Limits on Auckland Council	49
		42
	Auckland water organisations	
95	New sections 57A and 57B inserted	51
	57A Auckland water organisation must repay debt to Auckland Council	51
	57B Repeal of section 57A	52
96	Section 58 amended (Auckland water organisation must give to LTP and act consistently with other specified plans and strategies of Council)	effect 52
97	New section 60A inserted (Charges as security)	52
	60A Charges as security	52
98	Section 92 amended (Substantive council-controlled organisat must give effect to LTP and act consistently with other specifi plans and strategies of Council)	
	Consequential amendments	
99	Consequential amendments relating to Watercare Services Lin	nited 53
	Schedule 1	54
	Transitional, savings, and related provisions	
	Schedule 2	55
	Consequential amendments relating to Watercare Service Limited	ces

The Parliament of New Zealand enacts as follows:

1 Title

This Act is the Local Government (Water Services Preliminary Arrangements) Act **2024**.

Local Government (Water Services Preliminary Arrangements) Bill

2	Com	mencement			
(1)	This	This Act comes into force on the day after Royal assent.			
(2)		ever, sections 82 to 99 and Schedule 2 come into force on a date or set by Order in Council made on the recommendation of the Minister.			
(3)	-	provision of this Act that has not come into force by 1 July 2025 comes force then.	5		
(4)		Order in Council made under this section is secondary legislation (see 3 of the Legislation Act 2019 for publication requirements).			
		Part 1			
		Preliminary provisions	10		
3	Purp	oose			
(1)	The purpose of this Act is to establish a framework for local government to manage and deliver water services.				
(2)	This	Act delivers that purpose by—			
	(a)	requiring territorial authorities to prepare water services delivery plans; and	15		
	(b)	providing for the Minister to assist or intervene if territorial authorities find it difficult to prepare a water services delivery plan; and			
	(c)	requiring territorial authorities to provide other specified foundational information in relation to delivering water services, for the purpose of supporting economic regulation; and	20		
	(d)	providing specific consultation and decision-making processes that territorial authorities may use to establish, join, or amend council-controlled organisations that will deliver water services; and			
	(e)	providing a financially sustainable model for Watercare to be financially separate from Auckland Council and an interim economic regulation regime for Watercare that is administered by a Crown monitor.	25		
4	Over	view			
(1)		Part provides for preliminary matters, including the purpose of this Act he definitions of terms and expressions used in this Act.	30		
(2)	Part	Part 2 has the following subparts:			
	(a)	subpart 1 , which requires territorial authorities to prepare and submit water services delivery plans:			
	(b)	subpart 2 , which relates to Ministerial powers in relation to water services delivery plans:	35		
	(c)	subpart 3, which sets out a framework for specified territorial author-			

ities to disclose additional foundational information for the purposes of

6

cl 2

Part 1 cl 5

economic regulation, and also includes provisions relating to information sharing and enforcement.

- (3) **Part 3** sets out consultation and decision-making processes that territorial authorities may use when establishing, joining, or amending a water services council-controlled organisation.
- (4) **Part 4** sets out specific requirements for Watercare to be financially separate from Auckland Council, including the requirement for the Minister to appoint a Crown monitor and the requirement to have a Watercare charter.

5 Interpretation

In this Act, unless the context otherwise requires,—

10

5

amend, in relation to a council-controlled organisation, means to amend—

- (a) the constitution of the organisation; or
- (b) any other rules or documents that constitute the organisation or govern its activities

Commission means the Commerce Commission established by section 8 of the Commerce Act 1986

15

council-controlled organisation has the meaning set out in section 6(1) of the Local Government Act 2002

Crown facilitator means a Crown facilitator for water services delivery plans appointed under **section 20**

20

Crown water services specialist means a Crown water services specialist appointed under section 23

department means the department, departmental agency, or interdepartmental venture (as named in Part 1, 2, or 4 of Schedule 2 of the Public Service Act 2020) that, with the authority of the Prime Minister, is responsible for the administration of this Act

25

drinking water supply has the meaning set out in section 9 of the Water Services Act 2021

financial year means a period of 12 months ending on 30 June

financially sustainable means, in relation to a territorial authority's delivery of 30 water services, that—

- (a) the revenue applied to the authority's delivery of those water services is sufficient to ensure the authority's long-term investment in delivering water services; and
- (b) the authority is financially able to meet all regulatory standards and 35 requirements for the authority's delivery of those water services

firefighting water supplies has the meaning set out in section 6 of the Fire and Emergency New Zealand Act 2017

government department means a department listed in Part 1	of Schedule 2 of
the Public Service Act 2020	

joint arrangement means an arrangement between 2 or more territorial authorities to deliver water services (*see* **section 9**)

joint service area means,—

- 5
- (a) in relation to a joint arrangement, the combined districts of the territorial authorities that are a party to the joint arrangement; or
- (b) in relation to a joint WSCCO, the combined districts of the territorial authorities that control the joint WSCCO

joint water services council-controlled organisation or **joint WSCCO** 10 means a water services council-controlled organisation—

- (a) controlled by 2 or more territorial authorities; or
- (b) in which 2 or more territorial authorities are shareholders

joint water services delivery plan or **joint plan** means a water services delivery plan relating to 2 or more territorial authorities

15

LGA2002 means the Local Government Act 2002

long-term plan has the meaning set out in section 5(1) of the LGA2002

Minister means the Minister of the Crown who, under the authority of a warrant or with the authority of the Prime Minister, is responsible for the administration of this Act

20

ministerial appointee means—

- (a) a Crown facilitator for water services delivery plans appointed under **section 20**; or
- (b) a Crown water services specialist appointed under section 23

Secretary means the Secretary for Local Government

25

stormwater network means the infrastructure and processes that—

- (a) are used to collect, treat, drain, reuse, or discharge stormwater in an urban area; and
- (b) are owned by, or operated by, for, or on behalf of a territorial authority, a council-controlled organisation, or a subsidiary of a council-controlled organisation

30

Taumata Arowai means Taumata Arowai—the Water Services Regulator established by section 8 of the Taumata Arowai—the Water Services Regulator Act 2020

territorial authority has the meaning set out in section 5(1) of the LGA2002 35 **urban development** has the meaning set out in section 10 of the Urban Development Act 2020

wastewater network means the infrastructure and processes that—

Part 1 cl 7

(a)	are used to collect, store, transmit through reticulation, treat, or discharge wastewater; and				
(b)	are owned by, or operated by, for, or on behalf of a territorial authority, a council-controlled organisation, or a subsidiary of a council-controlled organisation				
wate	r services means services in relation to a territorial authority's—				
(a)	water supply network:				
(b)	stormwater network:				
(c)	wastewater network	10			
wate	r services council-controlled organisation or WSCCO—				
(a)	means a council-controlled organisation that—				
	(i) delivers water services; or				
	(ii) provides goods or services that are incidental and related to, or consequential on, delivering water services; and	15			
(b)	includes a joint water services council-controlled organisation; but				
(c)	does not include a council-controlled organisation that provides goods or services other than those listed in paragraph (a)				
wate	r services delivery plan—				
(a)	means a water services delivery plan prepared under subpart 1 of Part 2 ; and	20			
(b)	includes a joint water services delivery plan				
wate	r supply network means the infrastructure and processes that—				
(a)	are used to provide firefighting water supplies or drinking water supply; and	25			
(b)	are owned by, or operated by, for, or on behalf of a territorial authority, a council-controlled organisation, or a subsidiary of a council-controlled organisation				
	ercare means Watercare Services Limited and includes any subsidiary of reare Services Limited.	30			

The transitional, savings, and related provisions (if any) set out in **Schedule 1**

35

6

7

Transitional, savings, and related provisions

have effect according to their terms.

Act binds the Crown

This Act binds the Crown.

Part 2

Water services delivery plans and foundational information disclosure requirements

Subpart 1—Water services delivery plans

Water services delivery plan: preparation and contents

5

8 Territorial authority must prepare water services delivery plan

- (1) Each territorial authority must prepare a water services delivery plan that—
 - (a) identifies the current state of the authority's water services; and
 - (b) demonstrates publicly its commitment to deliver water services in a way that—

10

- (i) ensures that the territorial authority will meet all relevant regulatory quality standards for its stormwater network, wastewater network, and water supply network; and
- (ii) is financially sustainable for the territorial authority; and
- (iii) ensures that the territorial authority will meet all drinking water 15 quality standards; and
- (iv) supports the territorial authority's housing growth and urban development, as specified in the territorial authority's long-term plan.
- (2) Subsection (1) is subject to section 9(3).

20

9 Territorial authorities may enter into arrangement to submit joint plan

(1) A territorial authority may, in the circumstances described in **subsection (2)**, enter into an arrangement with 1 or more other territorial authorities for the purpose of submitting a joint water services delivery plan in relation to delivering the water services in the joint service area covered by that arrangement.

25

(2) For the purposes of **subsection (1)**, the circumstances are that the territorial authorities anticipate or propose delivering water services through a joint arrangement, as set out in a joint water services delivery plan.

•

(3) If 2 or more territorial authorities enter into an arrangement to submit a joint plan, the territorial authorities are required to prepare only 1 water services delivery plan in respect of the joint service area.

30

- (4) A joint arrangement must relate to the delivery of—
 - (a) all water services for all of the territorial authorities; or
 - (b) all water services except for some or all services relating to all of the territorial authorities' stormwater networks; or

Part 2 cl 11

(c) all water services for some of the territorial authorities, and all water services except for some or all services relating to stormwater networks for the other territorial authorities.

Example

Territorial authorities A, B, and C join together to form a joint arrangement. The joint arrangement relates to the delivery of all water services for territorial authorities A and B, but territorial authority C joins the joint arrangement in relation to water services other than those services relating to its stormwater network. Territorial authority C will deliver its services related to stormwater networks independently, rather than through the joint arrangement.

10

5

Alternatively, territorial authority C may join the joint arrangement in relation to water services other than some of its services relating to its stormwater network, and choose to deliver the remaining stormwater services independently.

10 Water services to be covered by water services delivery plan

(1) A water services delivery plan must relate to all water services.

15

(2) To avoid doubt, if a joint arrangement relates to the delivery of all water services except some or all of those relating to 1 or more of the authorities' stormwater networks, the joint plan must relate to the delivery of all water services in the joint service area.

Example

20

Territorial authorities A, B, and C join together to form a joint arrangement. The joint arrangement relates to the delivery of all water services for territorial authorities A and B, but territorial authority C joins the joint arrangement in relation to water services other than those services relating to its stormwater network. The joint plan must contain the required information in relation to all water services for all 3 territorial authorities, including information about territorial authority C's delivery of services relating to its stormwater network.

25

11 Contents of water services delivery plan

(1) A territorial authority's water services delivery plan must contain the following information in relation to the water services delivered in the authority's district:

30

- (a) a description of the current state of the water services network:
- (b) a description of the current levels of water services provided:
- (c) a description of
 - the areas in the district that receive water services (including a description of any areas in the district that do not receive water services); and

35

- (ii) the water services infrastructure associated with providing for population growth and development capacity:
- (d) whether and to what extent water services comply with regulatory requirements:

40

(e)	details of the capital and operational expenditure required—		
	(i) to deliver the water services; and		
	(ii) to ensure that water services comply with regulatory requirements:		
(f)	financial projections for delivering water services over the period covered by the plan, including—	5	
	(i) the operating costs and revenue required to deliver water services; and		
	(ii) projected capital expenditure on water infrastructure; and		
	(iii) projected borrowing to deliver water services:	10	
(g)	an assessment of the current condition, lifespan, and value of the water services network:		
(h)	a description of the asset management approach being used, including capital, maintenance, and operational programmes for delivering water services:	15	
(i)	a description of any issues, constraints, and risks that impact on delivering water services:		
(j)	the anticipated or proposed model or arrangements for delivering water services (including whether the territorial authority is likely to enter into a joint arrangement under section 9 or will continue to deliver water services in its district alone):		
(k)	an explanation of how the revenue from, and delivery of, water services will be separated from the territorial authority's other functions and activities:		
(1)	a summary of any consultation undertaken as part of developing the information required to be included in the plan under paragraphs (j) and (k):	25	
(m)	an explanation of what the authority proposes to do to ensure that the delivery of water services will be financially sustainable by 30 June 2028:	30	
(n)	any other information prescribed in rules made by the Secretary under section 14 .		
	ater services delivery plan must also comply with any requirements pre- ed in rules made by the Secretary under section 14 .		
Add	itional requirements for joint water service delivery plans	35	
A joi	int water services delivery plan must contain the following:		
(a)	information that clearly identifies each territorial authority that is proposed to be a party to the joint arrangement:		
(b)	information as to whether the joint arrangement will deliver—		

(2)

12

(1)

Part 2 cl 13

(i)	all water services for all of the territorial authorities that are par-
	ties to the joint arrangement; or

- (ii) all water services except for services in relation to all of the territorial authorities' stormwater networks; or
- (iii) all water services for some of the territorial authorities, and all water services except for services in relation to stormwater networks for the other territorial authorities:
- (c) all of the information listed in **section 11**:
- (d) information on the likely form of the joint arrangement, including whether it is anticipated it will involve water services being delivered by—
 - (i) a joint WSCCO; or
 - (ii) a joint local government arrangement or joint arrangement under section 137 of the LGA2002; or
 - (iii) another organisation or arrangement that the territorial authorities 15 are considering.
- (2) A joint water services delivery plan may also contain the following information to the extent that the information is available when the plan is submitted to the Secretary under **section 16**:
 - (a) an indicative implementation plan and timeline for the delivery of water 20 services under the joint plan:
 - (b) further information about the joint arrangement, including the ownership structure, the governance structure, and the control and financial rights of each territorial authority in the joint arrangement.
- (3) For the purposes of **subsection (1)(c)**, a joint plan must contain the information required under **section 11** in relation to—
 - (a) each territorial authority that is a party to the joint arrangement; and
 - (b) all water services delivered in the joint service area (including services relating to each territorial authority's stormwater network).
- (4) **Subsection (1)(c)** applies to a territorial authority's delivery of water services relating to its stormwater network even if the delivery of those services is not part of the joint arrangement.
- 13 Period covered by water services delivery plan
- (1) A water services delivery plan must cover a period of not less than 10 consecutive financial years, starting with the 2024–25 financial year.
- (2) A water services delivery plan must provide the required information—
 - (a) in detail in relation to each of the first 3 financial years covered by the plan; and

13

5

10

15

20

25

30

Local Government (Water Services Preliminary Arrangements) Bill

(b) in outline in relation to each of the subsequent financial years covered by the plan.

14 Secretary may make rules in relation to water services delivery plans

- (1) The Secretary may make rules for 1 or more of the following purposes relating to water services delivery plans:
 - (a) specifying additional information that must be included in a plan:
 - (b) specifying the manner and form in which information must be included in a plan.
- (2) Without limiting subsection (1)(a), a rule made under subsection (1)(a) may require the inclusion of information that—
 - (a) improves accountability to a territorial authority's community:
 - (b) provides a basis for regulating the delivery of water services:
 - (c) relates to 1 or more of the following:
 - (i) financial matters (including, for example, revenues, equity levels, debt arrangements, and expenses):
 - (ii) the assets involved in delivering water services (including, for example, asset management plans and asset replacement policies):
 - (iii) financial and non-financial performance measures:
 - (iv) the relevant performance measures and statistics relating to water quality:
 - (v) pricing practices, assumptions, policies and methodologies used in delivering water services.
- (3) Before making a rule, the Secretary must consult each person or organisation that the Secretary considers to hold views that are representative of the views held, or that may be held, in the local government sector.
- (4) A rule is not invalid only because the consultation required under **subsection** (3) occurred before this Act came into force.
- (5) Rules made under this section are secondary legislation (*see* Part 3 of the Legislation Act 2019 for publication requirements).

15 Process to prepare and adopt water services delivery plan

- (1) A territorial authority must adopt a water services delivery plan by resolution.
- (2) Except as provided in **Part 3** of this Act, a territorial authority must comply with subpart 1 of Part 6 of the LGA2002 (Planning and decision-making) when preparing, adopting, or amending a water services delivery plan.
- (3) This Act does not require a territorial authority to consult in relation to a water services delivery plan, but another enactment (for example, the LGA2002) may require a territorial authority to consult.

Part 2 cl 17

10

Submission of water services delivery plan

Plan must be submitted to Secretary for acceptance

(1)		A territorial authority or a joint arrangement (as the case may be) must submit its water services delivery plan to the Secretary—			
	(a)	no later than 1 year after the date on which this Act comes into force; or	5		
	(b)	on or before a later date specified by the Minister under section 17 .			

- (2) Each water services delivery plan that is submitted to the Secretary for acceptance must include a certification that—
 - (a) the plan complies with this Act; and
 - (b) the information contained in the plan is true and accurate.
- (3) The certification must be made,—

16

- (a) in the case of a joint water services delivery plan, by the chief executive of each authority to which the plan relates, in respect of the information provided by that authority; and
- (b) in any other case, by the chief executive of the territorial authority to 15 which the plan relates.

17 Minister may grant extension to deadline for submitting water services delivery plan

- (1) A territorial authority or a joint arrangement (the **applicant**) may apply to the Minister for an extension to the deadline for submitting its water services delivery plan.
- (2) An application for an extension must—
 - (a) be in writing; and
 - (b) specify the length of the extension that the applicant is seeking; and
 - (c) include sufficient information to enable the Minister to decide whether 25 to grant the extension; and
 - (d) be made no later than 1 month before the last date for submitting a plan under **section 16(1)(a)**.
- (3) The Minister may grant an extension only if the Minister is satisfied that—
 - (a) an application for the extension is made in accordance with **subsection** 30 **(2)**; and
 - (b) the applicant requires the extension for 1 or more of the following reasons:
 - (i) the applicant anticipates forming or joining a joint arrangement, and requires the extension to consult its communities in relation to the plan:

Part 2 cl 18

Local Government (Water Services Preliminary Arrangements) Bill

		(ii)	the applicant anticipates forming or joining a joint arrangement, and requires the extension to conclude the negotiations relating to forming or joining the joint arrangement:		
		(iii)	the applicant is late preparing its water services delivery plan as a result of having attempted, unsuccessfully, to form or join a joint arrangement:	5	
		(iv)	any other reason that the Minister considers justifies granting the exemption.		
(4)	Subs	sectio	n (3) is subject to subsection (5).		
(5)	for a		er may grant an extension despite not having received an application asion if satisfied that exceptional circumstances justify granting the	10	
(6)	notif	If the Minister grants an extension, the Minister must respond to the applicant, notifying it of the length of the extension and specifying the date by which the water services delivery plan must be submitted to the Secretary.			
(7)	The Minister may grant an extension subject to any conditions that the Minister thinks reasonable in the circumstances.				
			Acceptance of water services delivery plan		
18	Secr	etary a	accepts water services delivery plan		
(1)	The Secretary must—				
	(a)	consi and	der each water services delivery plan submitted under section 16 ;		
	(b)		ot a water services delivery plan only if satisfied that the plan comwith this Act.		
(2)	In deciding whether to accept a water services delivery plan, the Secretary may consult 1 or more of the following:				
	(a)	a gov	vernment department:		
	(b)	the C	Commission:		
	(c)	Taum	nata Arowai:		
	(d)	Crow	n Infrastructure Partners Limited.	30	
(3)	If the Secretary is not satisfied that a plan complies with the requirements in this Act, the Secretary must—				
	(a)	not s arran	the territorial authority or joint arrangement why the Secretary is satisfied with the plan and require the territorial authority or joint gement to amend the plan (which may be by including additional mation) and resubmit it to the Secretary by a specified date; or	35	
	(b)	decid	le not to accept the plan.		

Local Government (Water	Services	Preliminary
Arrangemen	ıts) Bill	

Part	2	cl	20

			Arrangements) bin 1 att 2 ct 20	
(4)			al authority or joint arrangement must comply with a requirement to resubmit a plan by the date specified.	
(5)			ling whether to accept a water services delivery plan, the Secretary the territorial authority or joint arrangement—	
	(a)	whet	ther the Secretary has accepted the plan; and	5
	(b)	if the	e Secretary has decided not to accept the plan, the reason for that sion.	
19	Publ	icatio	n of accepted water services delivery plan	
			etary notifies a territorial authority or joint arrangement that its water livery plan has been accepted,—	10
	(a)		erritorial authority or joint arrangement must, as soon as reasonably ticable, publish the water services delivery plan,—	
		(i)	in the case of a plan relating to 1 territorial authority, on the territorial authority's internet site; or	
		(ii)	in the case of a joint plan, on the internet site of each territorial authority to which the plan relates; and	15
	(b)		Secretary must, at the same time as the Secretary notifies having pted the plan, provide a copy of the plan to—	
		(i)	the Commission; and	
		(ii)	Taumata Arowai.	20
S	ubpar	t 2—	Ministerial powers in relation to water services delivery plans	
		C	rown facilitator for water services delivery plans	
20	Mini	ister n	nay appoint Crown facilitator	
(1)			ter may, in the circumstances set out in subsection (2) , appoint a litator for water services delivery plans to—	25
	(a)	a ter	ritorial authority; or	
	(b)	a gro	oup of 2 or more territorial authorities that is proposing to submit a	

the territorial authority or the group of territorial authorities requests, in writing to the Minister, that the Minister do so and the Minister decides

the Minister believes, on reasonable grounds, that it would be beneficial

joint water services delivery plan.

to grant that request; or

The Minister may appoint a Crown facilitator if—

to appoint a Crown facilitator because—

35

30

17

(2)

Local Government (Water Services Preliminary

121		Arrangements) Bill	
	(i)	the territorial authority or the group is otherwise unlikely to submit its plan to the Secretary in accordance with this subpart; or	
	(ii)	in the case of a group of territorial authorities, the group is having difficulty agreeing on the terms of a joint plan.	
		er may appoint either 1 person or a panel of 2 or more persons to be cilitator.	5
		ster appoints a panel to be a Crown facilitator, the Minister must number as the chairperson.	
How	Crow	n facilitator appointed	
The I	Minist	er appoints a Crown facilitator by—	10
(a)	facili	iding notice in writing to the person appointed to be the Crown stator or, if the Minister appoints a panel, to each member of the l; and	
(b)	appo	iding notice in writing to the territorial authority or, in the case of an intment to a group of territorial authorities, to each authority that is mber of the group; and	15
(c)	givin	g notice of the appointment in the Gazette.	
A no	tice un	der subsection (1) must include the following information:	
(a)	the to	erms of reference of the Crown facilitator, including—	
	(i)	an outline of the role the Crown facilitator has been appointed to undertake; and	20
	(ii)	the extent of the Crown facilitator's authority:	
(b)		ame of the Crown facilitator or, if the Crown facilitator is a panel, ame of each member of the panel:	
(c)	the s	tart and end dates of the Crown facilitator's appointment:	25
(d)	if the	e Crown facilitator is a panel, the name of the chairperson of the l:	
(e)		ame of each territorial authority to which the Crown facilitator has appointed.	
		er must notify any change in the membership of a Crown facilitator o each territorial authority to which the Crown facilitator is appoin-	30
Role	of Cr	own facilitator	

22 Role

- (1) A Crown facilitator for water services delivery plans may be appointed to do 1 or more of the following:
 - (a) assist the relevant territorial authority or group of territorial authorities to prepare a water services delivery plan:

18

(3)

Part 2 cl 21

(3)

(4)

21

(1)

(2)

Local Government (Wate	r Services	Preliminary
Arrangeme	nts) Bill	

Part 2 cl 23

(b)	advise the relevant territorial authority or group of territorial authorities how to prepare a water services delivery plan:	
(c)	assist the relevant territorial authority or group of territorial authorities to amend a draft water services delivery plan after being advised to do so by the Secretary (see section 18(3)(a)):	5
(d)	direct the relevant territorial authority or group of territorial authorities how to do 1 or both of the following:	
	(i) prepare a water services delivery plan:	
	(ii) if the Secretary has required a water services delivery plan to be amended (<i>see</i> section 18(3)(a)), amend a water services delivery plan:	10
(e)	assist 2 or more territorial authorities to agree on the terms of a joint arrangement, including, for example, by co-ordinating the negotiation process or by determining the terms of the joint arrangement:	
(f)	assist a territorial authority or group of territorial authorities to comply with the requirements in this Act:	15
(g)	anything else specified in the Crown facilitator's terms of reference.	
joint	ever, a Crown facilitator may be appointed to determine the terms of a arrangement in accordance with subsection (1)(e) only if the relevant orial authorities agree to the Crown facilitator having that role.	20
	art of performing its role, a Crown facilitator may also recommend to the ster that the Minister should take further action, for example, by—	
(a)	appointing a Crown water services specialist; or	
(b)	taking further action under Part 10 of the LGA2002.	
	Crown water services specialist	25
Mini	ster may appoint Crown water services specialist	
	Minister may, in the circumstances set out in subsection (2) , appoint a vn water services specialist to—	
(a)	a territorial authority; or	
(b)	a group of 2 or more territorial authorities that is proposing to form a joint arrangement.	30
The I	Minister may appoint a Crown water services specialist if—	
(a)	the territorial authority or the group of territorial authorities requests, in writing to the Minister, that the Minister do so and the Minister decides to grant that request; or	35
(b)	the territorial authority or the group has failed to submit its water ser-	

vices delivery plan to the Secretary in accordance with section 16; or

the territorial authority or the group—

19

(2)

(3)

23 (1)

(2)

(c)

		i) has submitted a water services delivery plan to the Secretary; but					
		despite the Secretary requiring the plan to be amended (see section 18(3)(a)), the plan does not comply with subpart 1; or					
	(d)	a Crown facilitator for water services delivery plans appointed to the ter- itorial authority or the group of territorial authorities has recommended 5 hat the Minister should make such an appointment, and the Minister agrees to do so.					
(3)		inister may appoint 1 person or a panel of 2 or more persons to be a water services specialist.					
(4)		Minister appoints a panel to be a Crown water services specialist, the r must appoint 1 member as the chairperson.	О				
24	How	rown water services specialist is appointed					
(1)	The N	nister appoints a Crown water services specialist by—					
	(a)	providing notice in writing to the person appointed to be the Crown water services specialist or, if the Minister appoints a panel, to each 1: member of the panel; and	5				
	(b)	providing in writing to the territorial authority or, in the case of an appointment to a group of territorial authorities, to each authority that is a member of the group; and					
	(c)	giving notice of the appointment in the <i>Gazette</i> .	0				
(2)	A notice under subsection (1) must include the following information:						
	(a)	he terms of reference of the Crown water services specialist, includ- ng—					
		i) an outline of the role the Crown water services specialist has been appointed to undertake; and	5				
		ii) the extent of the Crown water services specialist's authority:					
	(b)	he name of the Crown water services specialist or, if the Crown water services specialist is a panel, the name of each member of the panel:					
	(c)	he start and end dates of the Crown water services specialist's appointment:	0				
	(d)	f the Crown water services specialist is a panel, the name of the chair- person of the panel:					
	(e)	he name of each territorial authority to which the Crown water services specialist has been appointed.					
(3)		inister must notify any change in the membership of a Crown water ser- pecialist—	5				
	(a)	n writing to each territorial authority to which the Crown water services specialist is appointed; and					

Part 2 cl 27

(b) by notice in the *Gazette*.

25 Role of Crown water services specialist

- (1) A Crown water services specialist may be appointed to do 1 or more of the following:
 - (a) prepare a water services delivery plan for the territorial authority or the 5 group of territorial authorities:
 - (b) direct the territorial authority or the group of territorial authorities to adopt a specified water services delivery plan (which may be a plan that the specialist has prepared):
 - (c) direct the territorial authority or the group of territorial authorities to 10 submit a specified water services delivery plan to the Secretary under **section 16**:
 - (d) anything else specified in the specialist's terms of reference.
- (2) As part of performing its role, a Crown water services specialist may also recommend to the Minister that the Minister should take further action, for 15 example, taking further action under Part 10 of the LGA2002.

26 Decisions and directions of Crown water services specialist

- (1) This section applies to a territorial authority after the expiry of the term of a Crown water services specialist appointed to the territorial authority.
- (2) A direction given to the territorial authority by the Crown water services specialist ceases to have effect despite **section 27(c)**.
- (3) Despite **subsection** (2), any decision made by the territorial authority giving effect to a direction continues in force unless and until the territorial authority revokes or amends the decision.
- (4) A territorial authority that revokes or amends a decision under **subsection (3)** must, as soon as reasonably practicable, notify the Secretary for Local Government that the authority has done so.

Compare: 2002 No 84 s 258ZA(1)-(3)

General provisions

27 Obligation to co-operate

30

A territorial authority or a group of territorial authorities to which a ministerial appointee has been appointed must, as soon as reasonably practicable,—

- (a) co-operate with the ministerial appointee; and
- (b) comply with any reasonable request from the ministerial appointee to provide any relevant information that the territorial authority or the 3: group of territorial authorities holds; and

10

(c)	comply	with	any	reasonable	direction	issued	by	the	ministerial
	appointe	e.							

28 Minister retains powers under Part 10 of LGA2002

- (1) The Minister's powers under this subpart are in addition to, not in place of, the Minister's powers under Part 10 (Powers of Minister to act in relation to local authorities) of the LGA2002.
- (2) For the purpose of applying Part 10 of the LGA2002 in relation to a matter that arises under this Act, the definition of **problem** in section 256 of the LGA2002 must be read as including a failure by a territorial authority or group of territorial authorities to do 1 or more of the following:
 - (a) submit a water services delivery plan to the Secretary within the time frame specified in **section 16(1)**:
 - (b) have a water services delivery plan accepted by the Secretary within a reasonable period after submitting it for acceptance:
 - (c) give effect to proposals or undertakings specified in an accepted water 15 services delivery plan relating to the future delivery of water services:
 - (d) comply with section 27 of this Act.
- (3) In subsection (2)(c), an accepted water services delivery plan means a plan that—
 - (a) relates to the territorial authority or group of territorial authorities; and 20
 - (b) has been accepted by the Secretary under **section 18**.

29 Remuneration and expenses of ministerial appointee

- (1) A ministerial appointee is entitled—
 - (a) to receive remuneration for services as a Crown facilitator or as a Crown water services specialist (as applicable) as determined by the Minister in 25 accordance with the fees framework; and
 - (b) to be reimbursed for actual and reasonable travelling and other expenses incurred in carrying out their office as a Crown facilitator or as a Crown water services specialist in accordance with the fees framework.
- (2) In this section, **fees framework** means the framework determined by the Government from time to time for the classification and remuneration of statutory and other bodies in which the Crown has an interest.

Compare: 2002 No 84 s 258V

30 Recovery of expenses and remuneration from local authority

(1) A territorial authority owes as a debt to the Crown any remuneration and sexpenses that the Crown incurs for the appointment of a ministerial appointee to the territorial authority (whether individually or as a group of territorial

Part 2 cl 33

authorities), including the paymen	t of remuneration	and expenses	to the minis-
terial appointee.			

(2) The Crown may recover remuneration and expenses under **subsection (1)** as a debt to the Crown.

Compare: 2002 No 84 s 258W(1), (3)

5

31 Protection from liability

(1) A ministerial appointee is not liable for any act done or omitted to be done by them in good faith in the performance or intended performance of their functions, responsibilities, and duties, or the exercise of their powers, as a ministerial appointee.

10

(2) **Subsection (1)** applies, without limitation, to acts done or omitted to be done when directing a territorial authority or a group of territorial authorities.

Compare: 2002 No 84 s 258Y

Subpart 3—Foundational information disclosure requirements

32 Purpose of this subpart

15

- (1) The purpose of this subpart is to promote the long-term benefit of consumers of water services provided by territorial authorities.
- (2) This subpart achieves that purpose by promoting outcomes that are consistent with outcomes produced in competitive markets such that territorial authorities—

20

- (a) have incentives to-
 - (i) innovate and to invest in water services, including in replacement, upgraded, and new assets; and
 - (ii) improve efficiency in providing water services; and
 - (iii) provide water services at a quality that reflects consumer demands; and

25

- (b) share with consumers the benefits of efficiency gains in supplying water services, including through lower prices; and
- (c) are limited in their ability to extract excessive profits.

33 Application of this subpart

30

- (1) This subpart applies to 1 or more of the following entities that have been specified by the Governor-General by Order in Council made on the recommendation of the Minister and the Minister of Commerce and Consumer Affairs:
 - (a) a territorial authority that delivers water services:
 - (b) a council-controlled organisation that delivers water services:

35

(c) a subsidiary of a council-controlled organisation that delivers water services.

(2)		re making a recommendation under subsection (1) , the Minister and the ster of Commerce and Consumer Affairs must—	
	(a)	consider advice from the Secretary and the Commission; and	
	(b)	having considered that advice, believe that the entity to be specified in the Order in Council holds information that, if disclosed, would enable an interested person to assess whether the purpose of this subpart is being met.	5
(3)	An C matic	Order in Council made under this section must include the following infor- on:	
	(a)	the name of the entity:	10
	(b)	the water services to which a determination made under section 35 may apply.	
(4)		Order in Council made under this section is secondary legislation (see 3 of the Legislation Act 2019 for publication requirements).	
34	Mea	ning of specified entity	15
		is subpart, specified entity means an entity that is specified in an Order in a nicil made under section 33 .	
		Determinations	
35	Com	mission may make determination	
(1)	speci	Commission may make a determination setting out the information that a ified entity must make publicly available and must disclose to the Comion (see section 36).	20
(2)	The ation	Commission must consult interested parties before making a determin-	
(3)	A de	termination may relate to all specified entities or to 1 or more specified ies.	25
(4)	all w	not necessary for a single determination to address all matters relating to atter services and different parts of any determination may come into effect fferent times.	
(5)		etermination may require a specified entity to comply with the require- s set out in any other determination that has been made under this section.	30
(6)	Com	Commission may amend a determination in a material way only after the mission has consulted interested parties, but may amend a determination non-material way without prior consultation.	
(7)	sion	oon as practicable after making or amending a determination, the Commismust give to each specified entity to whom the determination relates to of the determination or the amendment (as applicable) and where it is able.	35

Part 2 cl 37

(8) A determination made under this section is secondary legislation (*see* Part 3 of the Legislation Act 2019 for publication requirements).

Compare: 2023 No 54 s 15

36 Effect of determination

- (1) A specified entity to which a determination made under **section 35** applies must—
 - (a) publicly disclose information in accordance with the requirements set out in the determination; and
 - (b) supply to the Commission a copy of all information disclosed in accordance with the determination within 5 working days after the specified 10 entity first makes the information available to the public.
- (2) The Commission—
 - (a) may monitor and analyse all information disclosed in accordance with this subpart; and
 - (b) must, as soon as practicable after any information is publicly disclosed, publish (on an internet site operated by or on behalf of the Commission) a summary and an analysis of that information for the purpose of promoting greater understanding of the performance of individual specified entities, their relative performance, and changes in their performance over time.
- (3) The Commission may, as part of a summary and an analysis, include an analysis of how effective the information disclosure requirements imposed on specified entities are in promoting the purpose in **section 32**.
- (4) In complying with **subsection (2)(b)**, the Commission must ensure that satisfactory provision exists to protect the confidentiality of any information that may reasonably be regarded as confidential or commercially sensitive.

Compare: 2023 No 54 s 34

37 Contents of determination

- (1) A determination made under **section 35** must specify the following:
 - (a) the specified entity to which it applies:

30

15

20

25

- (b) the water services to which it applies:
- (c) any time frames that must be complied with or that apply:
- (d) the information that the specified entity must disclose:
- (e) the manner in which the information must be disclosed:
- (f) the form of disclosure:

35

(g) when, and for how long, the specified entity must disclose the information:

10

15

20

- (h) any other methodologies that the specified entity must apply in preparing or compiling the information.
- (2) In making a determination under **section 35**, the Commission may have regard to the scale, complexity, and risk profile of each specified entity (or class of specified entity) to which the determination will apply (for example, by requiring more or less information to be disclosed).
- (3) A determination may require a specified entity to disclose information that includes, without limitation, 1 or more of the following:
 - (a) financial statements (including projected financial statements):
 - (b) asset values and valuation reports:
 - (c) prices, terms and conditions relating to prices, and pricing methodolo-
 - (d) contracts:
 - (e) transactions with related parties:
 - (f) financial and non-financial performance measures:
 - (g) plans and forecasts, including (without limitation) plans and forecasts about demand, investments, prices, revenues, quality and service levels, capacity and spare capacity, and efficiency improvements:
 - (h) asset management plans:
 - (i) quality performance measures and statistics:
 - (j) assumptions, policies, and methodologies used or applied in the delivery of water services:
 - (k) consolidated information that includes information about goods or services that are not incidental to, or related to, water services, in which case section 38 applies:
 - (l) information about the financing of territorial authorities and water services council-controlled organisations that includes information about goods or services that are not incidental to, or related to, water services, in which case section 38 applies.
- (4) In addition, a determination may require a specified entity to disclose information about how the entity is supporting and enabling planning processes, growth, and housing and urban development and, in particular, the entity's level of responsiveness in relation to those issues.
- (5) A determination may do 1 or more of the following:
 - (a) require disclosed information, or information from which disclosed 35 information is derived (in whole or in part), to be verified by statutory declaration:
 - (b) require independent audits of disclosed information:

Part 2 cl 38

- (c) require the retention of data on which disclosed information is based, and associated documentation:
- (d) provide for transitional provisions:
- impose any other requirements that the Commission considers necessary or desirable to promote the purpose of this subpart.

Compare: 2023 No 54 s 35

38 Determination may require specified entity to provide additional information

- (1) The purpose of this section is to enable the Commission to monitor—
 - (a) compliance with requirements to disclose information under this subpart 10 in relation to delivering water services; and
 - (b) the ongoing capability of a specified entity to raise finance with respect to its delivery of water services by assessing the specified entity's overall financial position.
- (2) A determination made under **section 35** may require a specified entity to disclose information referred to in **subsection (3)** only to the extent required to enable the Commission to monitor 1 or both of the matters referred to in **subsection (1)**.
- (3) If a specified entity provides goods or services that are not incidental to or related to delivering water services (other goods or services), a determination may require the entity to disclose—
 - (a) consolidated financial statements, and any other information referred to in **section 37**, for all businesses (including those related to the supply of other goods or services) undertaken by that entity; and
 - (b) consolidated financial statements, and any other information referred to in section 37, for the supply of all other goods or services in aggregate;
 - (c) reconciliation of information provided under **paragraphs (a) and (b)** with information disclosed in accordance with information disclosure requirements applying to delivering water services; and
 - (d) information about the financing of—
 - (i) all businesses (including those related to the supply of other goods or services) undertaken by that entity; and
 - (ii) the supply of all goods and services (including other goods or services) provided by that entity.

Compare: 2023 No 54 s 36

35

Local Government (Water Services Preliminary Arrangements) Bill

39	Cha	rge fo	r providing copies to public		
(1)	A specified entity that is required, by a determination made under section 35 , to provide copies of statements and information to the public on request may charge for providing those copies.				
(2)	prov	iding t	must be no more than is reasonably required to recover the costs of hose copies. 16 No 5 s 53E; 2023 No 54 s 37	5	
40			l monitoring and investigation powers based on subpart 8 of Commerce Act 1986		
(1)			rpose of carrying out its functions and exercising its powers under ne Commission may do any of the following:	10	
	(a)	cons	rult any person the Commission considers may assist it:		
	(b)	inve	stigate any of the following:		
		(i)	how effectively and efficiently a specified entity is delivering water services:	15	
		(ii)	how any conditions relating to the quality of water services may be, or are being, fulfilled:		
	(c)	asset	nine, consider, or investigate any activity, cost, revenue, transfer, t valuation, circumstance, or event that is occurring or that has arred during the previous 7 years.	20	
(2)	unde	r the r	nission's powers under subsection (1) are in addition to its powers est of this Act and under section 98 of the Commerce Act 1986.		
	comp				
			Sharing of information		
41	Shai	ing of	information and documents between Commission and	25	

41 Sharing of information and documents between Commission and department

- (1) The Commission and the department may share information with each other if the provider of the information believes that sharing the information is for either of the following purposes:
 - understanding a territorial authority's intention and commitment to 30 (a) deliver water services in a way that is consistent with the purpose of a water services delivery plan (see section 8); or
 - ensuring that sufficient information is available to interested persons to (b) assess whether the purpose of this subpart is being met.
- (2) Any information received by the department or the Commission under this Act may only be used in connection with,
 - in the case of the department, the performance or exercise of its functions, duties, or powers under this Act; or

28

Part 2 cl 39

Part 2 cl 42

(b)	in the case of the Commission, the performance or exercise of its func-
	tions, duties, or powers under this Act or under the Commerce Act 1986.

- (3) The department or the Commission may share information under this section whether or not a request has been made.
- (4) The department and the Commission must ensure that appropriate protections 5 are or will be in place to maintain the confidentiality of information shared under this section.
- (5) The department and the Commission may share commercially sensitive information under this section.
- (6) This section applies despite anything to the contrary in any contract, deed, or 10 document.
- (7) The department or the Commission may share the information subject to any conditions they think are appropriate.

Pecuniary penalty orders

42 When High Court may make pecuniary penalty order

15

- (1) The High Court may, on the application of the Commission, order a person to pay to the Crown a pecuniary penalty if the court is satisfied that the person has—
 - (a) contravened an obligation to disclose information under this subpart; or
 - (b) contravened an obligation to disclose information in the form or within 20 the time required; or
 - (c) disclosed information under this subpart that is false or misleading; or
 - (d) attempted to contravene an obligation to disclose information under this subpart; or
 - (e) been involved in a contravention of an obligation to disclose information 25 under this subpart.
- (2) In **subsection (1)(e)**, a person has been **involved in a contravention** if the person—
 - (a) has aided, abetted, counselled, or procured the contravention; or
 - (b) has induced the contravention, or attempted to induce it, whether by 30 threats or promises or otherwise; or
 - (c) has been in any way, directly or indirectly, knowingly concerned in, or party to, the contravention; or
 - (d) has conspired with others to effect the contravention.
- (3) Proceedings under this section may be commenced at any time within 3 years 35 after the contravention occurred.

Compare: 2023 No 54 ss 83(1), 126

43

Local Government (Water Services Preliminary Arrangements) Bill

Maximum amount of pecuniary penalty

(1)		maximum amount of a pecuniary penalty imposed under section 42 is, in ct of each act or omission,—	
	(a)	\$500,000, in the case of an individual; or	
	(b)	\$5 million, in any other case.	5
(2)		termining the amount of pecuniary penalty, the court must have regard to levant matters, including—	
	(a)	the nature and extent of the contravention; and	
	(b)	the circumstances in which the contravention took place (including whether the contravention was intentional, inadvertent, or caused by negligence); and	10
	(c)	whether the person has previously been found by the court in proceedings under this subpart to have engaged in similar conduct.	
(3)	•	rson may not be liable to more than 1 pecuniary penalty in respect of the conduct.	15
	Compa	are: 2023 No 54 s 84(1)	
		Orders about information disclosure	
44	Orde	r requiring information disclosure requirement to be complied with	
(1)	entity	High Court may, on application by the Commission, order a specified to comply with an obligation that applies to the entity to disclose informunder this subpart.	20
(2)	which	rder under this section must specify the date by which, or period within h, the specified entity must comply with the requirement.	
	Compe		
		Offence	25
45	Offer	nce relating to requirement to disclose information	
(1)	A per	rson commits an offence if—	
	(a)	the person, knowing that water services are subject to an obligation to disclose information under this subpart, intentionally contravenes any requirement in relation to that obligation; or	30
	(b)	the person is subject to an order under section 44 and fails to comply with the order by the date, or within the period, specified.	
(2)	tion t lion,	rson who commits an offence under subsection (1) is liable on convicto a fine not exceeding \$200,000, in the case of an individual, or \$1 milin any other case. are: 2023 No 54 s 98	35

Part 2 cl 48

		Miscellaneous provisions	
46	App	lication of Part 7 of Commerce Act 1986 (Miscellaneous provisions)	
		he purposes of this subpart, the following provisions of the Commerce Act apply with any necessary modifications:	
		Powers relating to evidence	5
	(a)	section 98 (Commission may require person to supply information or documents or give evidence):	
	(b)	section 98A (power to search) as if the reference to regulation under Part 4 of the Commerce Act 1986 were a reference to secondary legislation made under this Act:	10
	(c)	section 98G (Commission may exercise powers notwithstanding other proceedings):	
	(d)	section 99 (powers of Commission to take evidence):	
		Offences and administrative provisions	
	(e)	section 100 (powers of Commission to prohibit disclosure of information, documents, and evidence):	15
	(f)	section 100A (Commission may state case for opinion of High Court):	
	(g)	section 103 (offences):	
	(h)	section 104 (determinations of Commission):	
	(i)	section 106 (proceedings privileged):	20
	(j)	section 106A (judicial notice):	
	(k)	section 109 (Commission may prescribe forms).	
		Amendment to Local Government Act 2002	
47	Ame	ndment to Local Government Act 2002	
	Sect	tion 48 amends the Local Government Act 2002.	25
48	Secti	on 255 amended (Application of this Part)	
	Repl	ace section 255(2) with:	
(2)	-	ite subsection (1), the Minister may exercise the powers in this Part in on to—	
	(a)	a local board, and, for that purpose, this Part applies, with any necessary modifications, as if a local board were a local authority; or	30
	(b)	a territorial authority, a group of territorial authorities, or a joint arrangement for the purposes of the Local Government (Water Services Preliminary Arrangements) Act 2024 , and, for those purposes, this Part applies, with any necessary modifications, as if the territorial authority,	35

the group of te	rritorial authori	ities, or t	he joint	arrangement	were a	local
authority.						

(3) In subsection (2), joint arrangement has the same meaning as in section 5 of the Local Government (Water Services Preliminary Arrangements) Act 2024.

Part 3

Establishing water services council-controlled organisations

Preliminary provisions

49 Purpose of this Part

The purpose of this Part is—

10

5

- (a) to set out alternative consultation and decision-making requirements that territorial authorities can use (in place of certain consultation and decision-making requirements set out in the LGA2002) when establishing, joining, or amending a water services council-controlled organisation (the alternative requirements) (see sections 50 to 54); and
- (b) to give territorial authorities greater flexibility in relation to WSCCOs by setting out additional powers to, or exemptions from, specific provisions in the LGA2002 (see sections 55 to 58).

Alternative requirements

50 Alternatives to requirements in Local Government Act 2002

20

25

15

- (1) If a territorial authority complies with an alternative requirement specified in **sections 51 to 54**, it need not comply with the corresponding requirement in the LGA2002.
- (2) However, except as specified in this Part, all other relevant requirements in the LGA2002 continue to apply. For example, the requirements in sections 77(1)(c), 81, and 82(2) of the LGA2002 continue to apply to a territorial authority whether or not it complies with the alternative requirements.
- (3) A territorial authority may decide to rely on none, any, or all of the alternative requirements set out in **sections 51 to 54**.
- (4) A territorial authority that does not rely on an alternative requirement must comply with the corresponding requirement in the LGA2002.
- (5) Section 76 of the LGA2002 does not apply to the extent that a territorial authority complies with an alternative requirement.

51 Alternative requirement: decision making

(1) This section applies if a territorial authority is deciding whether or not to establish, join, or amend a water services council-controlled organisation.

Local Government (Water	Services	Preliminary
Arrangemei	ıts) Bill	

Part 3 cl 52

(2)	In th		se of that decision-making process, the territorial authority—	
	(a)	mus vice:	t identify both of the following 2 options for delivering water sers:	
		(i)	remaining with the existing approach for delivering water services; and	5
		(ii)	joining, forming, or amending (as the case may be) the WSCCO; but	
	(b)	may	identify additional options for delivering water services; and	
	(c)	mus	t assess the advantages and disadvantages of all options identified.	
(3)		-	rpose of section 50(1) , the corresponding requirement for this sec- ection 77(1)(a) and (b) of the LGA2002.	10
52	Alte	rnativ	e requirement: consultation	
(1)			n applies if a territorial authority is deciding whether or not to estaba water services council-controlled organisation.	
(2)		CCO, 1	territorial authority decides whether or not to establish or join the the territorial authority is only required to undertake consultation	15
(3)			bsection (2) , a territorial authority may decide to undertake further in before making the decision.	
(4)			ding whether to undertake further consultation, a territorial authority regard to—	20
	(a)	the r	requirement in section 78(1) of the LGA2002; and	
	(b)	ence	extent to which the authority already knows the views and prefer- es of persons likely to be affected by, or to have an interest in, the sion; and	25
	(c)	from	nature and significance of the decision, including its likely impact in the perspective of the persons who will or may be affected by, or an interest in, the decision.	
(5)			on applies despite anything to the contrary in the authority's signifi- engagement policy adopted under section 76AA of the LGA2002.	30
(6)	claus	se 48 c	al authority that defers adopting its 2024–2034 long-term plan under of Schedule 1AA of the LGA2002 may, to satisfy the requirement to the decision under this section, combine—	
	(a)	the c	consultation under this section; and	
	(b)	the a	authority's consultation on its 2025–2034 long-term plan.	35

For the purpose of **section 50(1)**, the corresponding requirement for this section is in section 56(1) of the LGA2002.

33

(7)

53 Alternative requirement: consultation on amendment to long-	ւջ-ւշւու լ	piai
--	------------	------

- (1) If a territorial authority is required to amend its long-term plan for the purpose of establishing, joining, or amending a WSCCO, the authority is not required to consult on the amendment if the authority—
 - (a) has already consulted its community in relation to the proposal to establish, join, or amend (as the case may be) a WSCCO; and
 - (b) is satisfied that its community has a good understanding of the implications of the proposal; and
 - (c) is satisfied that it understands its community's views on the proposal.
- (2) This section applies despite anything to the contrary in the authority's significance and engagement policy adopted under section 76AA of the LGA2002.
- (3) For the purpose of **section 50(1)**, the corresponding requirements for this section are in sections 93(5) and 97(2)(b) of the LGA2002.

54 Alternative requirement: information requirements for consultation

- (1) When a territorial authority consults about whether or not to establish, join, or amend a WSCCO, the authority must make the following information publicly available:
 - (a) the proposal, an explanation of the proposal, and the reasons for the proposal:
 - (b) an analysis of the reasonably practicable options (including the proposal), which must,—
 - if the authority relies on the alternative requirement in section 51(2), be at least the options identified under section 51(2)(a) and (b); or
 - (ii) in all other cases, be the options identified under section 77(1) of 25 the LGA2002:
 - (c) the likely consequences of proceeding with the proposal on the authority's rates, debt, and levels of service:
 - (d) the likely consequences of not proceeding with the proposal on the authority's rates, debt, and levels of service: 30
 - (e) if the proposal involves establishing, joining, or amending a joint WSCCO, the implications for communities throughout the joint service area of the joint WSCCO:
 - (f) if the proposal involves transferring ownership or control of a strategic asset to the WSCCO, a description of any accountability or monitoring arrangements the authority will use to assess the performance of the WSCCO in regard to the asset:
 - (g) any other relevant implications of the proposal that the authority considers will be of interest to the public.

Part 3 cl 56

- (2) For the purpose of **section 50(1)**, the corresponding requirement for this section is in section 82A(2) of the LGA2002.
- (3) In this section, strategic asset has the meaning set out in section 5(1) of the LGA2002.

Additional powers and exemption

5

55 Ability to consider joint service area

When deciding whether or not to establish, join, or amend a joint WSCCO, a territorial authority may, despite sections 12(4) and 14(1)(g) of the LGA2002, also consider—

- (a) the impact of the joint WSCCO on the communities in the joint service 10 area (as well as the impact on the authority's district); and
- (b) the views of people in communities in the joint service area (as well as the views of people in the authority's communities); and
- (c) the views of the other territorial authorities who are parties to the joint WSCCO.15

56 Joint committees

(1) Two or more territorial authorities that are considering whether or not to establish or amend a joint WSCCO may use a joint committee appointed under clause 30(1)(b) of Schedule 7 of the LGA2002 to perform 1 or more of the following tasks:

20

- (a) identify and assess the options under **section 51(2)** of this Act or section 77(1) of the LGA2002:
- (b) recommend a proposal to the territorial authorities for the purposes of consultation:
- (c) if the territorial authorities have authorised the joint committee to do so, 25 undertake consultation on behalf of the territorial authorities:
- (d) following all required consultation, recommend a decision to the territorial authorities.
- This section applies in addition to, and without limiting, any provision in Schedule 7 of the LGA2002 that relates to joint committees.
- (3) For the purposes of a joint committee performing any of the tasks listed in subsection (1), a reference in Part 6 of the LGA2002 (planning, decision-making, and accountability) to a local authority may be read as a reference to the joint committee.
- (4) If 3 or more territorial authorities are considering whether or not to establish or amend a joint WSCCO, some (but not all) of the territorial authorities may use a joint committee to perform the tasks listed in **subsection (1)**.

57	Ability to conditionally approve amending long-term plan	
(1)	This section applies if, for the purpose of establishing, joining, or amending a joint WSCCO, a territorial authority is required to—	
	(a) amend its long-term plan; or	
	(b) provide for the joint WSCCO when adopting its long-term plan.	5
(2)	The authority may approve the amendment or the adoption conditional on the other territorial authorities that are to be parties to the joint WSCCO—	
	(a) approving corresponding amendments to their long-term plans; or	
	(b) adopting their long-term plans with corresponding provisions.	
58	Exemption from cost-effectiveness review	10
(1)	This section applies if a territorial authority—	
	(a) is deciding whether or not to establish, join, or amend a WSCCO; or	
	(b) has established, joined, or amended a WSCCO.	
(2)	The authority is not required to undertake a review under section 17A of the LGA2002 in relation to the WSCCO.	15
59	Repeal of section 58	
	Section 58 and this section are repealed on the date that is 5 years after this Act comes into force.	
	Part 4	
	Watercare Services Limited	20
	Crown monitor	
60	Minister may appoint Crown monitor	
(1)	The Minister may appoint one of the following to be a Crown monitor to Watercare:	
	(a) an individual; or	25
	(b) a Crown entity; or	
	(c) a company named in Schedule 4A of the Public Finance Act 1989.	
(2)	In this section, Crown entity has the meaning set out in section 7(1) of the Crown Entities Act 2004.	
61	How Crown monitor appointed	30
(1)	The Minister must appoint a Crown monitor by providing notice in writing to—	
	(a) the Crown monitor; and	
	(b) Watercare.	

Part 4 cl 63

(2)	The	notice must include the following information:	
	(a)	the name of the Crown monitor; and	
	(b)	the date on which the Crown monitor's appointment starts.	
(3)	The	Minister may, by notice in writing to Watercare and the Crown monitor,—	
	(a)	change the terms of the Crown monitor's appointment; or	5
	(b)	end the Crown monitor's appointment.	
(4)	Crov	department must, as soon as practicable after the Minister appoints a vn monitor, changes the terms of an appointment, or ends an appointment, public notification of the Minister having done so—	
	(a)	on an internet site maintained by, or on behalf of, the department; and	10
	(b)	in a format that is readily accessible.	
62	Role	of Crown monitor	
(1)	The	role of the Crown monitor is to—	
	(a)	prepare a charter for Watercare (see section 63); and	
	(b)	review, and provide comments on, Watercare's business plan (see section 67); and	15
	(c)	monitor, and report on, Watercare's performance against the charter (see sections 71 and 72); and	
	(d)	take action to address any failure by Watercare to comply with the charter (see sections 76 to 81).	20
(2)	The	Crown monitor may specify—	
	(a)	the form and content of the business plan that Watercare must submit under section 65 :	
	(b)	the order in which Watercare must submit each component of a business plan.	25
		Watercare charter	
63	Wat	ercare charter	
(1)		rown monitor must prepare and make a Watercare charter.	
(2)		arter must comprise the following 2 parts:	
	(a)	Part 1 of the charter must contain (see section 64)—	30
		(i) minimum service quality standards for Watercare (which may include the time frame during which Watercare must meet the standards); and	
		(ii) financial performance objectives for Watercare; and	
		(iii) a customer compensation scheme for Watercare:	35
	(b)	Part 2 of the charter must—	

		(i)	contain a price-quality path for Watercare (see section 68); and	
		(ii)	specify the time period during which the charter applies (which must start no earlier than the day after the date on which the Crown monitor makes Part 2 of the charter).	
(3)	The	Crown	monitor must make—	5
	(a)	the 2	parts of the charter separately; and	
	(b)	Part 2	2 of the charter after it has made Part 1 of the charter.	
(4)			ring each part of a Watercare charter, the Crown monitor must con- owing:	
	(a)	Wate	rcare:	10
	(b)	Auck	cland Council:	
	(c)	Taun	nata Arowai:	
	(d)	the C	Commerce Commission.	
(5)			nade under this section is secondary legislation (see Part 3 of the Act 2019 for publication requirements).	15
			Part 1 of Watercare charter	
64	Con	tent of	Part 1 of Watercare charter	
	Mini	mum se	ervice quality standards	
(1)			service quality standards contained in Part 1 of a Watercare charter to 1 or more of the following:	20
	(a)	servi	ces provided by Watercare to consumers:	
	(b)	the p	erformance of Watercare's water supply network:	
	(c)	the p	erformance of Watercare's wastewater network:	
	(d)	the d	elivery of Watercare's capital investment.	
	Fina	ncial p	erformance objectives	25
(2)			erformance objectives contained in Part 1 of a Watercare charter e 1 or more of the following:	
	(a)		naximum amount of revenue that Watercare may earn on water supervices and wastewater services:	
	(b)		pproach that Watercare must use to recover the cost of its infrastructrough infrastructure growth charges:	30
	(c)	effici	ency targets that Watercare must achieve:	
	(d)	the m	ninimum credit rating that Watercare must maintain.	
	Cust	omer c	ompensation scheme	
(3)			compensation scheme contained in Part 1 of a Watercare charter by the compensation that Watercare must pay to a customer if Water-	35

Part 4 cl 66

care fails to meet a minimum service quality standard set out in the charter relating to that customer.

65	Watercare must	submit	business	plan	to	Crown	monitor
----	----------------	--------	----------	------	----	-------	---------

- (1) Watercare must submit a draft business plan to the Crown monitor.
- (2) Watercare must submit the draft business plan no later than 4 months after the date on which the Crown monitor makes Part 1 of the Watercare charter.
- (3) A business plan must, for the period during which it applies, include—
 - (a) the sources of, and Watercare's intended approach to, funding, revenue, and pricing; and
 - (b) Watercare's water infrastructure growth charging policy; and 10
 - (c) Watercare's intended approach to pricing its services and charging customers; and
 - (d) Watercare's financial strategy for each financial year covered by the plan; and
 - (e) Watercare's intended efficiency improvements for operating and capital 15 expenditure; and
 - (f) Watercare's investment priorities for its infrastructure assets; and
 - (g) how Watercare will—
 - (i) operate, maintain, and renew its infrastructure assets; and
 - (ii) provide new infrastructure assets; and

20

25

30

- (h) information about how the plan helps to achieve Watercare's proposed activities and intentions (as set out in its statement of intent).
- (4) Watercare must ensure that—
 - (a) the business plan that it submits complies with any requirements specified by the Crown monitor as to the form and content of the plan; and
 - (b) it provides each component of the plan to the Crown monitor in any order specified by the Crown monitor.

66 Period covered by business plan submitted to Crown monitor

- (1) A business plan that Watercare submits under **section 65** must cover a period of at least 10 consecutive financial years.
- (2) The Crown monitor may require that a business plan includes the required information—
 - (a) in detail in relation to each of the first 3 financial years covered by the plan; and
 - (b) in outline in relation to each of the subsequent financial years covered by 35 the plan.

67	Process	for	finalising	business	plan
----	---------	-----	------------	-----------------	------

- (1) After receiving a draft business plan under **section 65**, the Crown monitor—
 - (a) must review the plan; and
 - (b) may require Watercare to provide additional information relating to the plan; and
 - (c) may provide written comments on the plan to Watercare; and
 - (d) must specify a time frame for Watercare to submit a final version of the plan.
- (2) Watercare must submit a final version of the business plan after—
 - (a) providing any additional information requested by the Crown monitor; 10 and
 - (b) giving effect to any comments made by the Crown monitor.
- (3) Watercare must submit the final version of the business plan within the time frame specified by the Crown monitor.

Part 2 of Watercare charter

15

5

68 Content of Part 2 of Watercare charter

- (1) A price-quality path for Watercare contained in Part 2 of a Watercare charter (a **price-quality path**) must include the information required by the Crown monitor, which may be 1 or more of the following:
 - (a) the period to which it applies (which must not be more than 5 years): 20
 - (b) the minimum or maximum price or prices that Watercare may charge:
 - (c) the minimum or maximum revenue that Watercare may recover:
 - (d) the minimum service quality standards, performance targets, or financial performance objectives that Watercare must meet.
- (2) A price-quality path may include incentives for Watercare to maintain or 25 improve its services.
- (3) For the purposes of **subsection (2)**, the incentives may include (without limitation) any of the following:
 - (a) penalties by way of a reduction in Watercare's maximum prices or revenues based on whether, or by what amount, Watercare fails to meet the minimum service quality standards, performance targets, or financial performance objectives specified in Part 1 of the Watercare charter:
 - (b) rewards by way of an increase in Watercare's maximum prices or revenue based on whether, or by what amount, Watercare meets or exceeds the minimum service quality standards, performance targets, or financial performance objectives specified in Part 1 of the Watercare charter.

Part 4 cl 69

(a)	requi	irements to adopt a particular approach to risk management:
(b)	requi	irements in relation to the condition of assets and remaining asset
(c)	requi	irements to make particular types of investment:
(d)		irements to provide information about any investments planned for a cular period:
(e)		irements to consult the Crown monitor about certain kinds of invests and investment decisions:
(f)	requi	irements to adopt asset management policies and practices:
(g)	requi purp	irements to ring-fence minimum amounts of revenue for investment oses:
(h)	repoi	rting requirements, including—
	(i)	to whom reports must be made; and
	(ii)	the timing of reports; and
	(iii)	special reporting requirements in asset management plans, if Watercare fails to meet minimum service quality standards or per- formance targets; and
	(iv)	any other matters relating to reporting, including requirements for additional information:
(i)	whic ation	irements that any disclosed information, or any information from h disclosed information is derived, be verified by statutory declar- or certified (in the form specified by the Crown monitor) as true accurate:
(j)	-	irements to undertake cost-benefit analysis before Watercare begins specified projects:
(k)	requi	irements relating to consultation and engagement with consumers:
()		

Effect of Watercare charter

69 Effect of charter

the Crown monitor.

(1) After the Crown monitor makes Part 2 of the Watercare charter, the charter is binding on Watercare during the time period to which it applies.

(2)	If there is any inconsistency between obligations in the charter and obligations in Auckland Council's long-term plan, the obligations in the charter prevail.					
(3)	Each agreement for services entered into between Watercare and a customer of Watercare during the time period to which the charter applies must include any information relating to a customer compensation scheme that the charter requires.					
		Information disclosure				
70	Cro	wn monitor may require information disclosure				
(1)	info	Crown monitor may, by notice in writing, require Watercare to provide any mation the Crown monitor considers may enable the Crown monitor to orm or exercise its duties, functions, or powers under this Act.	10			
(2)		rmation that the Crown monitor may require Watercare to disclose may ide (without limitation) 1 or more of the following:				
	(a)	financial statements (including projected financial statements):				
	(b)	asset values and valuation reports:	15			
	(c)	prices, terms and conditions relating to prices, and pricing methodologies:				
	(d)	contracts:				
	(e)	transactions with related parties:				
	(f)	financial and non-financial performance measures:	20			
	(g)	plans and forecasts, including (without limitation) plans and forecasts about demand, investments, prices, revenues, quality and service levels, capacity and spare capacity, and efficiency improvements:				
	(h)	asset management plans:				
	(i)	quality performance measures and statistics:	25			
	(j)	assumptions, policies, and methodologies used or applied in relation to relevant information, including in relation to the information listed in paragraphs (a) to (i) .				
(3)	qual	the purpose of monitoring whether Watercare is complying with a price- ity path in Part 2 of the Watercare charter, the Crown monitor may, by the in writing to Watercare, require it to provide 1 or more of the following:	30			
	(a)	a written statement advising whether Watercare is complying with the price-quality path:				
	(b)	a written report analysing the written statement under paragraph (a), signed by—	35			

42

(i)

(ii)

Local Government (Water Service	ces Preliminary
Arrangements) Bil	11

Part 4 cl 73

	(c)	care is complying with the price-quality path.				
(4)	As part of requiring Watercare to disclose information under this section, the Crown monitor may require Watercare to provide a certificate confirming that the information it provides is true and accurate.					
(5)	A ce	rtificate must be—				
	(a)	in the form specified by the Crown monitor; and				
	(b)	signed by at least 1 director of Watercare.				
		Crown monitor to monitor and report on performance				
71	Crov	wn monitor must monitor Watercare's performance	10			
(1)	The	Crown monitor must monitor Watercare's performance under the charter.				
(2)		the purposes of subsection (1) , the Crown monitor is entitled to attend meeting of the board of Watercare.				
72	Crov	wn monitor must make annual report				
(1)	No later than 30 November in each year, the Crown monitor must report on Watercare's performance in the previous financial year against the following (contained in the Watercare charter):					
	(a)	minimum service quality standards or performance targets:				
	(b)	financial performance objectives:				
	(c)	the price-quality path.	20			
(2)	The	Crown monitor must—				
	(a)	provide the report to—				
		(i) Auckland Council; and				
		(ii) the Minister; and				
		(iii) the Minister of Commerce and Consumer Affairs; and	25			
	(b)	as soon as reasonably practicable after complying with paragraph (a) , make the report available to the public on an internet site maintained by, or on behalf of, the Crown monitor in a format that is readily accessible.				
73	Crov	wn monitor must make quarterly reports to Minister				
(1)	The	Crown monitor must provide a quarterly report to the Minister.	30			
(2)	A quarterly report must report on the Crown monitor's performance or exercise of its functions, duties, or powers under this Act.					

Crown monitor's expenses

74	Crow	n monitor's expenses are recoverable from Watercare				
(1)	The Crown monitor's expenses in relation to monitoring Watercare must be reimbursed—					
	(a)	by Watercare to the Crown monitor; and	5			
	(b)	on any terms and conditions set by the Minister.				
(2)	The te	erms and conditions may, for example, do 1 or more of the following:				
	(a)	specify, or specify classes, descriptions, or kinds of, all or any of the expenses:				
	(b)	impose a cap on classes of expenses or total expenses:	10			
	(c)	specify a time period in which classes of, or total, expenses are incurred, for the purposes of calculating a cap.				
(3)		uty to reimburse the expenses is not the Crown lending money for the ses of the Public Finance Act 1989.				
	Compa	re: 2022 No 77 Schedule 1 cl 35	15			
		Commerce Commission's functions and powers				
75	Comr	nerce Commission's functions and powers				
(1)		the purposes of this Act, the Commission may review Parts 1 and 2 of a Watercare charter and provide comments to the Crown monitor.				
(2)		urpose of the Commission's review and provision of comments is to help imise—	20			
	(a)	the charter; and				
	(b)	the application of the charter to Watercare; and				
	(c)	decisions made by the Crown monitor in relation to the charter.				
(3)		rt of its function and powers under this Act, the Commission may engage my party it considers practicable (for example, Taumata Arowai).	25			
(4)	If the	Minister appoints the Commission to be the Crown monitor,—				
	(a)	the Commission has the functions, powers, and duties required to perform that role; but				
	(b)	section 63(4)(d) and this section do not apply.	30			
		Enforcement				
76	High	Court may impose orders				
(1)		figh Court may, on application by the Crown monitor, make 1 or more of ders listed in subsection (2) if the court is satisfied that Watercare—				
	(a)	has contravened the Watercare charter; or	35			

(b)

Local Government (Water Services Preliminary Arrangements) Bill

has attempted to contravene the Watercare charter.

Part 4 cl 77

(2)	The orders are as follows:							
	(a)	an order requiring Watercare to comply with the charter:						
	(b)	an injunction restraining Watercare from contravening the charter:						
	(c)	an order requiring Watercare to pay to the Crown a pecuniary penalty:	5					
	(d)	any other order that the court considers appropriate in the circumstances, including an order directing Watercare to pay to the Crown the costs of the proceedings.						
(3)	In th	is section, contravening the charter includes—						
	(a)	failing to comply with the requirements in a price-quality path, whether by charging a price for services that is higher than the maximum price permitted, or by receiving more revenue than is permitted, or in any other way:	10					
	(b)	failing to comply with the requirements in any minimum service quality standards or performance targets, or in any financial performance objectives:	15					
	(c)	failing to comply with the requirements relating to a customer compensation scheme.						
(4)		eedings under this section are not able to be commenced later than 3 years the contravention occurred.	20					
77	Max	imum amount of pecuniary penalty						
(1)		maximum amount of a pecuniary penalty imposed under section 76 is 000,000 in respect of each act or omission.						
(2)		etermining the amount of pecuniary penalty, the court must have regard to elevant matters, including—	25					
	(a)	the nature and extent of the contravention; and						
	(b)	the nature and extent of any loss or damage suffered by any person as a result of the contravention; and						
	(c)	the circumstances in which the contravention took place (including whether the contravention was intentional, inadvertent, or caused by negligence); and	30					
	(d)	whether the court, in proceedings under this Act or any other legislation, has previously found Watercare to have engaged in any similar conduct.						
(3)		ercare may not be liable to more than 1 pecuniary penalty in respect of the conduct.	35					

78 Order to disclose information

- (1) The High Court may, on application by the Crown monitor, make 1 or more of the orders listed in **subsection (2)** if the court is satisfied that Watercare has failed—
 - (a) to comply with a notice under **section 70** requiring Watercare to disclose information; or
 - (b) to comply with the requirement to submit a business plan in accordance with section 65; or
 - (c) to address the Crown monitor's comments on a draft business plan, as required under **section 67(2)(b)**.
- (2) The orders are as follows:
 - (a) an order directing Watercare to comply with the relevant obligation:
 - (b) an order requiring Watercare to pay to the Crown a pecuniary penalty:
 - (c) any other order that the court considers appropriate in the circumstances, including an order directing Watercare to pay to the Crown the costs of the proceedings.

79 Maximum amount of pecuniary penalty

- (1) The maximum amount of a pecuniary penalty imposed under **section 78** is \$300,000 in respect of each act or omission.
- (2) Watercare may not be liable to more than 1 pecuniary penalty in respect of the same conduct.

80 Further penalty for continuing breach

- (1) For a continuing breach, the High Court may impose, for each day or part of a day during which the breach continues, a further penalty in addition to a pecuniary penalty imposed under **section 76 or 78**.
- (2) The further penalty under subsection (1) is—
 - (a) \$500,000 for a breach referred to in **section 76**; and
 - (b) \$50,000 for a breach referred to in **section 78**.
- (3) A further penalty under **subsection (1)** may be imposed only in respect of the period that—
 - (a) begins on the day on which the pecuniary penalty was imposed under **section 76 or 78** (as applicable); and
 - (b) ends on the day on which the breach is remedied.

81 Appeal

(1) A party to proceedings under **section 76** who is dissatisfied with an order or a decision of the High Court under that section may, with the leave of the Court

46

Local Government (Water	Services	Preliminary
Arrangemer	ıts) Bill	

Part 4 cl 87

of Appeal,	appeal	to t	hat	court	on	a	question	of	law	against	the	order	or	deci-
sion.														

- (2) In determining an appeal under this section, the Court of Appeal may exercise any power of the High Court in respect of proceedings under **section 76**.
- (3) An appeal must be made by giving notice of appeal
 - a) not later than 20 working days after the date on which the order was made or notice of the decision was communicated to the appellant; or
 - (b) within any further time that the Court of Appeal allows.

Related amendments to LGA2002

82 Principal Act

10

5

Sections 83 to 91 amend the Local Government Act 2002.

83 Section 121 amended (The Crown not liable for debts)

Replace section 121(1) with:

 The Crown is not liable to contribute to the payment of any debts or liabilities of—

15

- (a) any local authority; or
- (b) Watercare Services Limited.

84 Section 122 amended (Disclosure document and loan documents to contain statement that the Crown does not guarantee financial products or loan)

In section 122(1) and (3), after "local authority", insert "or Watercare Services Limited".

20

85 Section 124 amended (Interpretation)

In section 124, insert in their appropriate alphabetical order:

Auckland has the meaning set out in section 4(1) of the Local Government (Auckland Council) Act 2009

25

Watercare means Watercare Services Limited, and includes any subsidiary of Watercare Services Limited

86 Section 127 amended (Duty to ensure communities have access to drinking water if existing suppliers facing significant problems)

After section 127(3)(a), insert:

30

(aa) a **territorial authority** (despite the definition of that term in section 5(1)) includes Watercare:

87 Section 130 amended (Obligation to maintain water services)

(1) In the heading to section 130, after "water services", insert ": general".

After section 130(4), insert:

(2)

Local Government (Water Services Preliminary Arrangements) Bill

(5)			ion, local government organisation has the meaning given in seccept it—			
	(a)		des Auckland Council only in relation to its provision of storm- r services in Auckland; and	5		
	(b)	exclu	ides Watercare.			
88		section rcare)	n 130A inserted (Obligation to maintain water services:			
	After	sectio	n 130, insert:			
130A	Oblig	gation	to maintain water services: Watercare	10		
(1)			must continue to provide water services in Auckland and maintain to meet its obligations under this subpart.			
(2)	In ord	der to	fulfil the obligations under this subpart, Watercare must—			
	(a)	not u	se assets of its water services as security for any purpose:			
	(b)	not d	ivest its ownership or other interest in a water service:	15		
	(c)	struc	ose control of, sell, or otherwise dispose of the significant infrature necessary for providing water services in Auckland, unless, in g so, it retains its capacity to meet its obligations:			
	(d)	not, i	n relation to a property to which it supplies water,—			
		(i)	restrict the water supply unless section 193 applies; or	20		
		(ii)	stop the water supply unless section 25 of the Water Services Act 2021 applies.			
(3)	This	section	does not override sections 131 to 137.			
(4)	In thi	s secti	on, water services means water supply and wastewater services.			
89	Section	on 253	3 amended (Outline of Part)	25		
	In sec	ction 2	53(a) and (b), after "local authorities", insert "or Watercare".			
90	Section	on 254	4 amended (How this Part works)			
(1)	Repla	ice sec	etion 254(2) with:			
(2)	Subpart 1 provides the Minister with a range of options in relation to a local authority or Watercare if they have a problem. The options are available only in relation to the local authority or Watercare itself, and not to any entity that the local authority or Watercare may control or have an interest in.					
(2)	In sec	ction 2	54(4) and (5), after "local authority", insert "or Watercare".			
91	Section	on 255	5 amended (Application of this Part)			
	After	sectio	on 255(2), insert:	35		

Part 4 cl 94

(3)Despite subsection (1), the Minister may exercise the powers in section 257 to 258E in relation to Watercare and, for that purpose, sections 256 to 258E, 258N to 258Q, and 258S to 258ZA apply as if Watercare were a local authority.

Related amendments to Local Government (Auckland Council) Act 2009

92 **Principal Act**

5

Sections 93 to 98 amend the Local Government (Auckland Council) Act 2009.

93 Section 4 amended (Interpretation)

In section 4(1), replace the definition of **Auckland water organisation** with:

Auckland water organisation means Watercare Services Limited, and includes any subsidiary of Watercare Services Limited

10

94 New section 56A and cross-headings inserted

After the Part 5 heading, insert:

Auckland Council

56A Limits on Auckland Council

15

- (1)The Auckland Council-
 - (a) has no right, title, or interest (legal or equitable) in the assets, security, debts, or liabilities of an Auckland water organisation; and
 - must not receive any equity return, directly or indirectly, from an Auck-(b) land water organisation; and

20

25

- must not give an Auckland water organisation any financial support or (c) capital; and
- must not lend money or provide credit to an Auckland water organisation; and
- must not give any person any guarantee, indemnity, or security in rela-(e) tion to the performance of any obligation by an Auckland water organisation: and
- must not direct an Auckland water organisation in relation to any borrowing of any sort by that organisation.
- (2) In this section,—

30

borrowing-

- means the incurring by any means of debt to raise money; and (a)
- (b) includes the incurring of debt
 - under any contract or arrangement for hire purchase, deferred pay-(i) ment, instalment payment, sale and lease-back or buy-back, finan-

35

		cial lease, loan, overdraft, or other arrangement for obtaining debt finance; or	
	(ii)	by the drawing, acceptance, making, endorsement, issue, or sale of bills of exchange, promissory notes, and other negotiable instruments and debt securities; or	5
	(iii)	by the use, for any purpose, of funds received or invested by the Auckland water organisation for any other purpose if the Auckland water organisation has resolved to repay, with or without interest, the funds used; or	
	(iv)	under any contract for services; but	10
(c)	good	not include debt incurred in connection with the hire purchase of ls, the deferred purchase of goods or services, or the giving of credit he purchase of goods or services if—	
	(i)	the period for which the indebtedness is outstanding is less than 91 days and the indebtedness is not incurred again promptly after payment; or	1:
	(ii)	the goods or services are obtained in the ordinary course of the Auckland water organisation's performance of its lawful responsibilities, on terms and conditions available generally to parties of equivalent creditworthiness, for amounts not exceeding in aggregate an amount—	20
		(A) determined by resolution of the Auckland water organisa- tion as not being so significant as to require specific author- isation; or	
		(B) recorded for the purposes of this subsubparagraph in the then current borrowing management policy of the Auckland water organisation; and	2:
(d)	does	not include a contract for services that is entered into—	
	(i)	in the ordinary course of the Auckland Council's or an Auckland water organisation's performance of its lawful responsibilities; and	30
	(ii)	on terms and conditions generally available to other parties of equivalent creditworthiness	
capi	tal incl	ludes uncalled capital	
equi	ty retu	ırn means—	3:
(a)	profi	ts of an Auckland water organisation; or	
(b)	distri	ibutions from an Auckland water organisation; or	
(c)	orgai	benefit derived, directly or indirectly, from an Auckland water nisation that represents, is calculated by reference to, or is deter-	40

Part 4 cl 95

	(1)	a share in or proportion of an Auckland water organisation's capital; or	
	(ii)	an Auckland water organisation's surplus or residual economic value (after satisfying prior contractual claims); or	
	(iii)	an Auckland water organisation's profitability or any other indicator of its success	5
		ial support or capital does not include to enter into any contract to sell or supply goods or services on credit—	
(a)		e ordinary course of the Auckland Council's, or an Auckland water isation's, performance of its lawful responsibilities; and	10
(b)		rms and conditions generally available to other parties of equivalent tworthiness	
lend 1	money	or provide credit—	
(a)	inclu	des, without limiting the generality of that expression,—	
	(i)	to defer payment for any goods or services supplied or works constructed for any person, organisation, or government; and	15
	(ii)	to enter into hire purchase agreements or agreements that are of the same or a substantially similar nature; and	
	(iii)	to enter into finance lease arrangements or arrangements that are of the same or a substantially similar nature; and	20
	(iv)	to subscribe for any debt securities or uncalled capital; but	
(b)		not include to enter into any contract for services to sell or supply s or services on credit—	
	(i)	in the ordinary course of the Auckland Council's, or an Auckland water organisation's, performance of its lawful responsibilities; and	25
	(ii)	on terms and conditions generally available to other parties of equivalent creditworthiness	
	ity has Act 20	s the meaning set out in section 6(1) of the Financial Markets Con- 13.	30
Compa	re: 2022	No 77 s 171	
		Auckland water organisations	
New	section	ns 57A and 57B inserted	

95

After section 57, insert:

57A Auckland water organisation must repay debt to Auckland Council

If, on the date on which this section comes into force, an Auckland water (1) organisation owes a debt to the Auckland Council in respect of water services

51

		structure, the Auckland water organisation must repay that debt, including nterest payable.	
(2)		Auckland water organisation must repay a debt under subsection (1) ite anything in section 56A .	
(3)	The	repayment—	5
	(a)	may be made by instalments; but	
	(b)	must be paid in full no later than the close of the day that is 5 years after this section comes into force.	
57B	Repe	eal of section 57A	
		section and section 57A are repealed on the date that is 5 years after this on comes into force.	10
96		ion 58 amended (Auckland water organisation must give effect to LTP act consistently with other specified plans and strategies of Council)	
	After	r section 58(3), insert:	
(4)	This	section is subject to—	1.
	(a)	section 69(2) (effect of charter) of the Local Government (Water Services Preliminary Arrangements) Act 2024 ; and	
	(b)	section 56A (limits on Auckland Council) of this Act.	
97	New	section 60A inserted (Charges as security)	
	After	section 60, insert:	20
60A	Cha	rges as security	
(1)		section applies if—	
	(a)	an Auckland water organisation has granted a security interest over a charge or charging regime revenue as security for a loan or the performance of any obligations under an incidental arrangement; and	25
	(b)	a receiver has been appointed under section 40A or 40B of the Receiverships Act 1993 in respect of that loan or arrangement; and	
	(c)	the Crown monitor has been informed of the appointment.	
(2)		receiver may, without further authority than this section, assess and collect ch financial year a charge under this section to recover sufficient funds to	30
	(a)	the payment of the Auckland water organisation's commitments in respect of the loan or incidental arrangement during that year; and	
	(b)	the reasonable costs of administering, assessing, and collecting the	
		charge.	3:

Part 4 cl 99

(3)		ever, a receiver may not create, or receive, any interest or security in water ces infrastructure.	
(4)		arge under this section must be assessed as a uniform charge in the dollar water services charges of a property—	
	(a)	in Auckland; or	5
	(b)	if the Auckland water organisation resolved that, at the time when the loan was being raised or the incidental arrangement was being entered into, it was for the benefit of only a specified area, that area.	
(5)		arge under this section over any 1 or more of the assets of an Auckland r organisation is subject to section $40D(5)$ and (6) of the Receiverships Act .	10
(6)	tion	is section, Crown monitor means a Crown monitor appointed under sec-60 of the Local Government (Water Services Preliminary Arrangements) 2024 .	
98	give	ion 92 amended (Substantive council-controlled organisations must effect to LTP and act consistently with other specified plans and regies of Council)	15
	Afte	section 92(2), insert:	
(3)	This	section is subject to—	
	(a)	section 69(2) (effect of charter) of the Local Government (Water Services Preliminary Arrangements) Act 2024 ; and	20
	(b)	section 56A (limits on Auckland Council) of this Act.	
		Consequential amendments	
99	Cons	sequential amendments relating to Watercare Services Limited	
		nd the legislation specified in Schedule 2 as set out in that schedule.	25

Schedule 1 Transitional, savings, and related provisions

s 6

Part 1 Provisions relating to this Act as enacted

5

There are no transitional, savings, or related provisions in this Act as enacted.

Schedule 2

Schedule 2

Consequential amendments relating to Watercare Services Limited

s 99

Part 1 Amendments to primary legislation

5

Civil Defence Emergency Management Act 2002 (2002 No 33)

In section 115A, after "local authority", insert "or Watercare Services Limited" in each place.

Local Government (Rating) Act 2002 (2002 No 6)

After section 19(2), insert:

10

- (3) **Subsection (1)** does not apply to Auckland Council.
- (4) In **subsection (3)**, **Auckland Council** means the local authority established by section 6(1) of the Local Government (Auckland Council) Act 2009.

Receiverships Act 1993 (1993 No 122)

In the cross-heading above section 40A, after "authorities", insert "and Watercare". 15 Before section 40A, insert:

40AAA Meaning of Watercare

In sections 40A to 40E, Watercare means Watercare Services Limited.

In section 40A, after "local authority", insert "or Watercare".

Replace section 40B with:

20

40B Power of court to appoint receiver

- (1) Subject to sections 40D and 40E and this section, the High Court may,—
 - (a) on the application of any creditor of a local authority, appoint a receiver of any asset of the local authority or appoint a receiver for the purposes of section 115 of the Local Government Act 2002:

25

- (b) on the application of any creditor of Watercare, appoint a receiver of any asset of Watercare or appoint a receiver for the purposes of section 60A of the Local Government (Auckland Council) Act 2009.
- (2) An appointment under subsection (1) must be for the period, and with the rights, powers, and duties, and on any terms and conditions, including as to security and remuneration, that the court considers appropriate in all the circumstances.
- (3) When considering, in accordance with subsection (2), the terms and conditions upon which a receiver can be appointed by a court in relation to a local authority, the court must—

35

55

Recei	ivershi	ps Act	t 1993 (1993 No 122)—continued	
	(a)		account of the interests of both the secured and non-secured credit- the local authority, as against—	
		(i)	the interests of the local authority itself; and	
		(ii)	the requirement of the local authority to provide those services that are essential for the maintenance of public health and safety; and	5
		(iii)	the interests of the ratepayers with property within the area of the local authority; and	
		(iv)	the interests of the general public living within the area of the local authority; and	10
	(b)		account of the interests of secured creditors as against the interests n-secured creditors of the local authority.	
(4)	tions	upon v	dering, in accordance with subsection (2) , the terms and condiwhich a receiver can be appointed by a court in relation to Waterart must—	15
	(a)		account of the interests of both the secured and non-secured credit- Watercare, as against—	
		(i)	the interests of Watercare itself; and	
		(ii)	the requirement of Watercare to provide those services that are essential for the maintenance of public health and safety; and	20
		(iii)	the interests of consumers with property in Auckland; and	
		(iv)	the interests of the general public living in Auckland; and	
	(b)		account of the interests of secured creditors as against the interests n-secured creditors of Watercare.	
(5)			on, Auckland has the same meaning as in section 4(1) of the Local t (Auckland Council) Act 2009.	25
ln sec	ction 40	D(1),	after "local authority", insert "or Watercare" in each place.	
n sec	ction 40)D(3),	after "local authority", insert "or Watercare".	
n sec	ction 40)D(4),	after "local authority's", insert "or Watercare's".	
Repla	ice sect	tion 40	D(5) with:	30
(5)			ubsection (6), subsection (5A) applies to any land that is vested thority or Watercare and is—	
	(a)	a rese	erve under the Reserves Act 1977; or	
	(b)	land of ition;	over which the local authority or Watercare has no power of disposor	35

land in respect of which the local authority's or Watercare's power of

56

(c)

disposition is conditional.

Schedule 2

Receiverships Act 1	993 (1993 N	No 122)—co	ntinued
---------------------	-------------	------------	---------

(5A) The power of disposition that a receiver of that local authority or Watercare has in respect of the land is limited to a power of disposition by way of lease or licence for a term or terms not exceeding in the aggregate 9 years.

In section 40D(6), after "local authority", insert "or Watercare".

In section 40E(1), (2), and (3)(a), after "local authority", insert "or Watercare".

5

Part 2

Amendments to secondary legislation

National Civil Defence Emergency Management Plan Order 2015 (LI 2015/140)

In the Schedule, heading to Part 10, after "local authorities", insert "and Watercare Services Limited".

10

In the Schedule, clause 159(1) and (4), after "local authorities", insert "and Watercare Services Limited".

In the Schedule, clause 160, after "local authorities", insert "or Watercare Services Limited".

In the Schedule, after clause 161(7), insert:

15

(8) In this clause, **local authority** has the meaning given in clause 2(2) but also includes Watercare Services Limited.

In the Schedule, heading to clause 162, after "local authorities", insert "and Watercare Services Limited".

In the Schedule, clause 162, delete "by local authorities to".

20

In the Schedule, clause 162(a), before "care", insert "by local authorities to".

In the Schedule, clause 162(b) and (c), before "take", insert "by local authorities and Watercare Services Limited to".

In the Schedule, heading to clause 163, after "local authorities", insert "and Watercare Services Limited".

25

In the Schedule, clause 163(1), after "local authorities", insert "or Watercare Services Limited".

Wellington, New Zealand:

Published under the authority of the New Zealand Government—2024

House of Representatives

Amendment Paper

Local Government (Water Services Preliminary Arrangements) Bill

Proposed amendment

Hon Simeon Brown, in Committee, to move the following amendment:

New Part 5

After clause 99 (page 53, after line 25), insert:

Part 5 Amendments to Water Services Act 2021

100 Principal Act

Sections 101 and 102 amend the Water Services Act 2021.

101 Section 138 amended (Wastewater environmental performance standards)

After section 138(3), insert:

- (3A) When making wastewater environmental performance standards under this section, Taumata Arowai must not have regard to the hierarchy of obligations in clause 1.3(5) of the National Policy Statement for Freshwater Management.
- (3B) **Subsection (3A)** applies despite any other provision of this Act.
- 102 New section 138A inserted (Repeal of provisions relating to National Policy Statement for Freshwater Management)

After section 138, insert:

Proposed amendments to

Local Government (Water Services Preliminary Arrangements) Bill

AP No 41

138A Repeal of provisions relating to National Policy Statement for Freshwater Management

- (1) The Governor-General may, by Order in Council, on the recommendation of the Minister, repeal section 138(3A) and (3B) and this section.
- (2) The Minister may make a recommendation under subsection (1) only when the Minister for the Environment—
 - (a) recommends the approval of a new national policy statement under section 52 of the Resource Management Act 1991 to replace the National Policy Statement for Freshwater Management; or
 - (b) reviews, changes, or revokes the National Policy Statement for Freshwater Management under section 53(1) of the Resource Management Act 1991.
- (3) An Order in Council made under this section is secondary legislation (see Part 3 of the Legislation Act 2019 for publication requirements).

Explanatory note

This Amendment Paper inserts *new Part 5* into the Local Government (Water Services Preliminary Arrangements) Bill (the **Bill**).

New Part 5 of the Bill makes 2 amendments to the Water Services Act 2021 (the **principal Act**).

The first amendment is to insert *new subsections (3A) and (3B)* into section 138 of the principal Act. Section 138 of the principal Act authorises Taumata Arowai to make wastewater environmental performance standards. Section 14 of the principal Act requires Taumata Arowai, when making those standards, to give effect to Te Mana o te Wai (as that term is defined in that Act) to the extent that it applies. *New section 138(3A) and (3B)* provides that, when making wastewater environmental performance standards, Taumata Arowai must not have regard to the hierarchy of obligations contained in Te Mana o te Wai.

The second amendment is to insert *new section 138A* into the principal Act. *New section 138A* authorises the Governor-General to make an Order in Council to repeal *new sections 138(3A)* and (3B) and 138A on the recommendation of the Minister responsible for the administration of the principal Act. The Minister may make a recommendation only when the Minister for the Environment recommends either replacing the National Policy Statement for Freshwater Management 2020 (under section 52 of the Resource Management Act 1991) or reviewing, changing, or revoking that national policy statement (under section 53(1) of the Resource Management Act 1991).

Proposed amendments to

Local Government (Water Services Preliminary Arrangements) Bill

AP No 41

Departmental disclosure statement

The Department of Internal Affairs is required to prepare a disclosure statement to assist with the scrutiny of this Amendment Paper. The disclosure statement provides access to information about any material policy changes to the Bill and identifies any new significant or unusual legislative features of the Bill as amended.

A copy of the statement can be found at http://legislation.govt.nz/disclosure.aspx?type=ap&subtype=government&year=2024&no=41&.

Regulatory impact statement

No regulatory impact statement (**RIS**) was prepared specifically for the matters in this Amendment Paper. However, a RIS was prepared on 3 April 2024 for related changes to the National Policy Statement for Freshwater Management 2020, which is available at https://environment.govt.nz/assets/publications/regulatory-impact-statement-for-freshwater-rm-amendment-bill.pdf

Wellington, New Zealand:
Published under the authority of the House of Representatives—2024



Hamilton City Council – Draft 1 Submission

Local Government (Water Services Preliminary Arrangements) Bill

Parliament's Finance and Expenditure Select Committee

13 June 2024





Improving the Wellbeing of Hamiltonians

Hamilton City Council is focused on improving the wellbeing of Hamiltonians through delivering to our five priorities of shaping:

- A city that's easy to live in
- A city where our people thrive
- A central city where our people love to be
- A fun city with lots to do
- A green city

The topic of this Council submission is aligned to all of Hamilton City Council's five priorities.

Council Approval and Reference

This Council submission was approved by Hamilton City Council at its extraordinary meeting that was held on 11 June 2024.

Submission # 767

Introduction

- Hamilton City Council welcomes the opportunity to make a submission to Parliament's Finance and Expenditure Select Committee on the Local Government (Water Services Preliminary Arrangements) Bill.
- **2.** This submission is structured under the following key areas:
 - Improving the Wellbeing of Hamiltonians
 - Part A General Comments
 - Part B Detailed Considerations
 - Specific Matters for Consideration
 - Further Information and Hearings

Improving the Wellbeing of Hamiltonians

- **3.** Hamilton City Council is the territorial authority that is responsible for providing the wellbeing of residents in New Zealand's fourth-largest city. It is the lawful provider of water services to its community and was the fastest-growing city in the country last year.
- **4.** Under the Local Government Act 2002, Hamilton City Council must fulfil its purpose to enable democratic local decision-making and action by, and on behalf of, communities; and to promote the social, economic, environmental, and cultural well-being of communities in the present and for the future.
- **5.** Cost-effective provision of high quality, sustainable water services is critical to all wellbeing outcomes for our community.
- **6.** Economically, Hamilton is one of the key drivers of the Waikato's economic success. Its proximity to two main seaports (Auckland and Tauranga), two international airports (Auckland and Hamilton), rail connections and distribution networks, the south Auckland industrial base and key state highways provide significant opportunities for trade.
- 7. The Waikato-Tainui Te Kauhanganui Incorporated is the principal constitutional and legally mandated local iwi authority, encompassing some 33 hapuu and 67 marae across several local authority boundaries. Waikato-Tainui takes on the wider governance focus for its people, its tribal culture, education, and social responsibility. Hamilton City Council and Waikato-Tainui work together to give effect to Te Ture Whaimana o Te Awa o Waikato the Strategy and Vision for the Waikato River.
- 8. It is in this context that Hamilton City Council provides its submission to this Select Committee on the Local Government (Water Services Preliminary Arrangements) Bill. The Bill establishes the Local Water Done Well framework and the preliminary arrangements for the new water services system.

Part A - General Comments

- 9. This submission is in two parts Part A being an overview of general comments which outline Council's position on Local Water Done Well and consideration of matters arising from the Local Government (Water Services Preliminary Arrangements) Bill, and Part B which considers specific clauses in the Bill.
- **10.** Hamilton City Council strongly supports reform of the national water services sector, acknowledging the current model is unsustainable for councils and our communities.
- **11.** We support this reform being led by Government, as successful reform is critical to New Zealand's public health, economic growth, housing provision and environmental sustainability.

- **12.** Our communities work, travel and live across territorial and regional boundaries. Our primary raw water sources and discharge points are not constrained by territorial boundaries. Impacts on water quality and availability of resource in one council area have flow-on impacts to others.
- **13.** Sustainable and cost-effective water services delivery is not a council issue, it is a national consideration.
- 14. Hamilton City Council recognises the benefits of aggregation of water services delivery across territorial authority boundaries. Almost a decade ago we invested millions of dollars in exploring a Council Controlled Organisation (CCO) delivery model which would have delivered \$480 million in efficiencies over 25 years if Hamilton, Waikato, and Waipa Councils were to operate water services as one network.
- **15.** Political considerations saw that effort fail.
- **16.** Hamilton ratepayers are now again faced with significant costs to explore this issue again, at the direction of Government.
- **17.** Local Water Done Well is a much-needed reform of this sector but as a New Zealand issue it must be supported by national funding. It is unfair to expect Hamilton ratepayers to fund the lion's share of any regional solution to a national issue.
- **18.** The previous Government's reform programme, and the direction indicated by this Government, is imposing greater regulatory and compliance costs on Council, at a time when ratepayers are already bearing the brunt of high inflation, high interest rates, increasing insurance costs and a cost-of-living crisis.
- **19.** The previous Government's reform identified significant costs for councils from transitioning water services. It offered funding to address the financial risks our Council faces in funding and managing an organisation without a third of our current activity.
- **20.** Our own analysis indicated this funding was \$20 million short for our city.
- **21.** The previous reform also centrally funded cross-council services through transition, such as digital services and workforce transition, reducing the per-ratepayer quantum through economies of scale.
- **22.** Local Water Done Well is yet to offer any provision to offset these costs, and in fact increases costs to this Council.
- **23.** The Local Government (Water Services Preliminary Arrangements) Bill and Government policy imposes the costs of the Water Services Delivery Plan, the costs of increased disclosure, the costs of paying for Commerce Commission oversight, and the costs for any intervention by the Minister.
- **24.** In our view, it is essential that in considering this Bill the Select Committee addresses the intent of the Regulatory Review Agency and the approach by this government to ensure any new requirements are absolutely necessary and are the most cost-effective approach to addressing any identified problems.
- **25.** The Bill requires us to prepare a Water Services Delivery Plan within 12 months of this Bill's enactment that identifies our current state, ensures we meet all compliance and standards requirements, supports housing growth and urban development, and is financially sustainable.
- **26.** This is a significant and costly task, yet we are advised by Government that key information we need to properly develop this plan will not be available until late in 2024, when a further Bill will detail options, tools, and models for councils.
- 27. In order to find the best outcomes for Hamilton we need to consider a range of possible future delivery options and potential partnerships with other councils. Yet at this stage we are still waiting for critical policy decisions and legislation that will shape what a Water Services Council-Controlled Organisation (WSCCO) looks like, their powers and authorities, and the way in which councils relate to them.

- **28.** Key issues include the statutory powers and authorities of a waters CCO, the ability to access property, control connections, make use of bylaws (or not), manage trade wastes, act as a requiring authority under the Resource Management Act and Public Works Act, the charging mechanisms that will be available to a WSCCO, and the taxation status of a WSCCO.
- **29.** These are material matters in relation the legal ability of a waters CCO to undertake functions currently undertaken by the Council. They are also material matters in relation to the cost-effectiveness of alternative delivery models.
- **30.** A change in the tax status of delivery of water services would increase costs significantly, with no benefit to ratepayers or water users.
- **31.** Legislative clarity is needed quickly in order to support the decisions that councils will need to make as they prepare their Water Services Delivery Plans. We urge the government to expedite resolution of these issues and provide clarity as soon as possible.
- **32.** Hamilton City Council provides high quality drinking water, wastewater, and stormwater services to more than 180,000 residents and thousands of visitors to Hamilton every day. We're committed to protecting our environment, supporting public health, and enabling sustainable growth, while meeting our regulatory responsibilities.
- **33.** Through Local Water Done Well, we will continue to seek the best (financial and non-financial) sustainable outcomes for the communities of Hamilton now and for the future.
- **34.** Any transition must prioritise the health of the awa and Hamilton's role in implementing Te Ture Whaimana and cater for Hamilton's growth needs.
- **35.** But this transition will come at a cost, and it is not a cost which should fall unreasonably and unfairly on the shoulders of Hamilton ratepayers.
- **36.** We urge this Government to recognise the national benefits of this reform and take appropriate steps to ensure the costs are equitably shared by all who will benefit from it.

Part B - Detailed Considerations

- **37.** In responding to Local Water Done Well and the considerable challenges that Hamilton faces Hamilton City Council is seeking to ensure that waters are delivered in a way that:
 - **1.** Provides the best (financial and non-financial) sustainable outcomes for the communities of Hamilton now and for the future.
 - 2. Supports the health of the awa and Hamilton's role in implementing Te Ture Whaimana.
 - 3. Responds to and caters for Hamilton's growth needs as identified in Hamilton's growth strategy.
 - **4.** Realises the benefits of a boundaryless approach across the wider metro area / Future Proof subregion, supporting the best land use, transport, development and sustainability outcomes.
 - **5.** Provides a stable and secure long term decision-making environment that can make sound investment decisions for very long-life infrastructure.
 - **6.** Ensures that all environmental, public health, and economic regulatory compliance requirements are met.
 - 7. Ensures commitments and obligations to iwi and hapū are met.
 - **8.** Creates the conditions necessary to build, attract, develop, and sustain the highly skilled workforce needed to deliver water services.
 - **9.** Ensures that Hamiltonians have influence in ownership and governance decisions proportionate to the size and nature of the City's population, assets, and needs.
 - **10.** Recognises Hamilton's role and contribution to the region and the benefits of regional action.

- **38.** Hamilton City Council also seeks to ensure that:
 - **1.** Change follows the most logical, efficient, cost effective, and lowest risk pathway to realise the benefits.
 - 2. The community clearly understands any change.
- **39.** Council considers that the Bill as introduced is a significant step towards the outcomes that it is seeking to achieve, but it is only a first step, and we urge the Government to progress at pace the next phase of its Local Water Done Well Reform. There are many fundamental questions that are still to be answered in relation to the tools that will be available to address the waters challenges that we face. Speedy clarification of the tools and policy environment for waters entities is crucial if we are to deliver the outcomes we seek.
- **40.** Council generally supports the Bill as introduced. We support the purpose of the Bill, the broad framework of the proposed Water Services Delivery Plans, the circumstances in which the Minister for Local Government may consider the appointment of a Crown facilitator or Crown waters specialist, and the step towards economic regulation through the Commerce Commission.
- **41.** There are a number of detailed matters where Hamilton City Council has concerns and seeks changes to the Bill. They are set out in the following table: **Specific Matters for Consideration**.
- **42.** One of the key issues that we have identified in the definition of the stormwater network. This is a significant technical issue with considerable consequences.
- **43.** Whilst Council supports the policy intent of being able to effectively integrate the management and funding of all three waters, we remain very concerned over the ability to sensible and practically define the stormwater network. The piped stormwater network is relatively easy to identify. But what surrounds that is considerably more difficult.
- **44.** Over years streams have been both piped as part of development and daylighted as part of more modern approaches to urban development and stormwater management. Which parts constitute the stormwater network as defined in the Bill?
- **45.** Council's parks and reserves are key parts of our communities' open space and recreation facilities; they also perform critical stormwater management functions. Roads are critical both in terms of their impact on generating stormwater runoff and it the connections that they provide to both the piped and un-piped flow of stormwater. Are they part of the stormwater network under the Bill's definitions?
- **46.** Most modern urban developments include significant stormwater detention facilities, wetlands, tanks, and dams. Some of these become Council property when the development is complete, others remain in private ownership. Functionally these are critical to the effective management of stormwater and flooding risks, but these are excluded from the definition of stormwater in the Bill.
- **47.** Our Council urges the Select Committee to carefully consider stormwater, the way stormwater networks are defined and the way in which stormwater fits within the regulatory framework. We submit that the management of stormwater is a fundamentally different issue than the provision of drinking water and the collection and treatment of wastewater. This needs to be reflected in the statutory framework.
- **48.** Hamilton City Council has not addressed the specifics of the provisions relating to Watercare, they are matters for Watercare, Auckland Council, and the people of Auckland to consider.
- **49.** However, we note there are aspects of the proposed Watercare approach that would be concerning if they were to become entrenched as the national model. In particular, we are concerned to ensure that any approach to the potential liability for debts does not result in a more difficult and more costly borrowing environment for any other new waters entity.

50. We also want to ensure that disclosure and monitoring is cost effective and not burdensome. We need to ensure that any changes to delivery arrangements deliver better outcomes – not more costly solutions for our ratepayers. Hamilton City Council is anxious to work with the Government to ensure that the next step in water reform does indeed deliver benefits for our communities.

Specific Matters for Consideration

- **51.** The following table sets out the specific matters that Hamilton City Council would like to see addressed in the Bill.
- **52.** Hamilton City Council would welcome the opportunity to talk with the Finance and Expenditure Select Committee about the matters raised in this submission.

Section of the Bill	Issue	Requested Change
5 - Interpretation	The definition of a stormwater network is	Develop a clearer and unambiguous
Stormwater	ambiguous and arguably includes roads	definition of stormwater networks
Network	and other Council infrastructure such as	and stormwater assets. This is
	parks and reserves which provide a	critical for both clarity of the scope
	stormwater function in addition to their	of a Water Services Delivery Plan,
	core purpose. Amongst other things,	the scope and impact of reporting
	roads, parks, and reserves are	requirements and the impact of
	infrastructure and processes that are used	economic regulation.
	to collect, treat, drain, reuse, or discharge	
	stormwater in an urban area; and are	
	owned by, or operated by, for, or on behalf	
	of a territorial authority.	
s8(1)(b)(i) to (iv)	In addition to the factors listed it is critical	Add to s8(1)(b) as follows (or
	that the communities served by a Water	similar):
	Services Delivery Plan have security	
	around the ability to take the amount of	(v) ensures security of water supply
	water necessary to meet their needs for	and security in the ability to
	drinking water, discharge the amount of	discharge from wastewater and
	treated wastewater effluent they will	stormwater networks.
	generate, and discharge stormwater into	
	the receiving environment. Financial	
	sustainability is not sufficient if there is no	
	security of supply or ability to discharge.	
	Security of supply and ability to discharge	
	depends on being able to secure consents	
	to take water and discharge into water	
	bodies. Both may be compromised by	
	over-allocation of water bodies. Security	
	of supply and ability to discharge should	
	be a key factor in the design of any	
	economic regulation and will be a major	
	driver of future costs.	
s8(1)(b)(iv)	A territorial authority's Long Term Plan	Amend ss(1)(iv) to refer to either a
	currently sets out only the financial	Future Development Strategy or to
	aspects of the territorial authority's plans	the growth and development
	to respond to housing and urban	provided for in the relevant District
	development pressures, and large	Plan.
	infrastructure investments. For growth	
	councils a more credible link may be to a	

Section of the Bill	Issue	Requested Change
	Future Development Strategy prepared	
	under the National Policy Statement for	
	Urban Development. Alternatively, a	
	reference could be made to the growth	
	and urban development provided for in	
	the Territorial Authority's District Plan.	
	Because a Future Development Strategy	
	relates to the whole of an urban area,	
	referring to it is a more helpful way of	
	ensuring that the investment necessary to	
	support growth is provided for.	
s11(e)	Two of the key challenges facing growth	Amend ss(e) by adding:
	councils are the very large capital and	
	operational expenditure that is needed to	and; (iii) to ensure infrastructure
	support growth and development, and the	required to support housing growth
	timing of growth-related capital works	and urban development can be
	needed to enable development. The scope	delivered:"
	of a Water Services Delivery Plan needs to	
	include a clear reference to these issues	
	so that there can be confidence in the	
	ability to service growth.	
s11(h)	The term "asset management approach"	Either clarify the term in the Bill or
	is ambiguous and not a term commonly	ensure that it is suitable defined and
	used in the sector	clear in any rules established by the
		Secretary.
s13(1)	The proposed period covered by a Water	Amend s13(1) so that the start year
	Services Delivery Plan starts in the 2024-	for Water Service Delivery Plans is
	25 financial year. The Plans are due to be	2025-26.
	adopted by Councils by 30 June 2025. By	
	that time the 2024-25 financial year will	
	be over.	
s14	The Bill imposes a new requirement on	Establish a deadline of 31 October
	territorial authorities. The preparation of a	2024 for the Secretary to make rules
	Water Services Delivery Plan will be a	in relation to Water Services Delivery
	new, unfunded requirement that imposes	Plans.
	further costs on ratepayers. Rules that	
	may be made by the Secretary have the	
	potential to add further costs, or to	
	impose requirements that have not been	
	anticipated by territorial authorities. Rules	
	may also require particular approaches	
	(for instance in cost allocation) that may	
	require considerable work (or re-work) by	
	territorial authorities in order to comply.	
	In order to minimise the impact on	
	ratepayers it is critical that any rules made	
	by the Secretary are specified early	
	enough for councils to respond to,	
	carefully plan work, and avoid any	
	omissions in, or rework of, draft plans.	
	Receiving a new set of requirements late	
	in the period in which plans must be	

Section of the Bill	Issue	Requested Change
	produced would add further costs and	
	risks to the preparation of Water Services	
	Delivery Plans.	
s14(3)	As noted above, the Bill imposes a new requirement on territorial authorities. The preparation of a Water Services Delivery Plan will be a new, unfunded requirement that imposes further costs on ratepayers. In keeping with the Government's commitment to reducing red tape and ensuring that regulations are fit for purpose it would be appropriate to include a requirement that any rules made by the Secretary are reasonable and that the cost of complying with the rules is outweighed by the benefits of implementing the rule. Given the potential impact of rules that can be made by the Secretary, the costs that they may impose on ratepayers, and the number of territorial authorities, it is not unreasonable to expect that the Secretary would be required to consult with those who will be impacted by the rules, not just with representatives who are identified by the Secretary. Hamilton	Amend s14(3) read: (3) Before making a rule, the Secretary must: (i) consult the organisations required to produce Water Services Delivery Plans and also consult any person or organisation that the Secretary considers to hold views that are representative of the views held, or may be held, in the local government Sector; and (ii) be satisfied that any additional costs imposed on territorial authorities through a rule are reasonable and are outweighed by the benefits of implementing the rule.
	City Council would expect to be consulted	
	about such key matters.	
S20 and s23	The appointment of a Crown facilitator or a Crown water services specialist may be a valuable step in meeting the purposes of the Bill, but such appointments are interventions in the legitimate democratically elected authority of territorial authorities and should not be taken lightly. A Crown water services specialist would have significant powers, akin to those of a Crown manager under the Local Government Act 2002. There would be significant costs associated with the work of either a Crown facilitator or a Crown water services specialist. Given the costs associated with such appointments, and the fact that these costs are borne not by the Minister, but by the territorial authorities to which an appointment is made, it is reasonable to expect that before making an appointment the Minister must be satisfied that the benefits of the appointment exceed the costs and that the expenditure represents value for money for the authorities who	Add to s20(2)(b) and (c) the Minister is satisfied that the benefits of making the appointment exceed the costs of the appointment. Add to s23(2) and (e) the Minister is satisfied that the benefits of making the appointment exceed the costs of the appointment.

Section of the Bill	Issue	Requested Change
	are paying for the appointments.	
s21 and s24	To avoid arguments, the appointment of a Crown facilitator, or Crown water services	Add to s21(2)
	specialist to a group of councils needs to include that way in which their charges will be allocated to each of the councils in the group. The allocation of costs should not be the subject of debate or argument between the authorities to which	(f) the way in which the costs of the Crown facilitator will be allocated between the territorial authorities to which the Crown facilitator has been appointed.
	appointments have been made.	and s24(2)
		(f) the way in which the costs of the Crown water services specialist will be allocated between the territorial authorities to which the Crown water services specialist has been appointed.
s35, s37(2)	Hamilton City Council generally supports the shift to economic regulation, provided it is cost-effective and does not impose a new and costly burden on ratepayers for	Add to s35 a new subsection and renumber existing ss (3) to (8) as necessary.
	little or no benefits. Hamilton City Council is particularly concerned to ensure that any additional regulation is fit for purpose, cost effective, and necessary.	new (3) The Commission must be satisfied that the benefits arising from the determination will exceed the costs of compliance.
		Add to s37(2)
		and must have regard to the costs that the determination will impose on the specified entities and be satisfied that the benefits arising from the determination will exceed the costs of compliance.
s37(3)(k) and (l), s37(4), and s38(3)	Hamilton City Council is particularly concerned that the scope of potential regulation by the Commerce Commission extends well beyond waters and across the broad sweep of territorial authority operations. Territorial authority reporting is already subject to considerable regulation and complex accounting and reporting standards. These sections add to the that. Section 37(3)(k) and (l), s37(4), and s38(3)	Amend the Bill to ensure that the ability of the Commerce Commission to make a determination that impacts on the nature, level and extent of information relating to non-waters activities of local authorities is strictly limited and provides no scope to increase the complexity of reporting, or duplicate reporting that may be required by the Secretary, or under other regulations.
	opens the door to de facto economic regulation of all local government activity. Hamilton City Council submits that this is beyond the purpose of the Bill, is a step too far, and is likely to be expensive and risk considerable duplicative effort by	This could include: • A requirement to consult the Secretary and ensure that any determination does not duplicate

Section of the Bill	Issue	Requested Change
	territorial authorities. These sections take the role of the Commerce Commission into the core reporting requirements of local authorities, which are already complex and heavily regulated.	 other requirements or add undue new requirements; Limitation on the scope of s38(3) in relation to non-waters activity; and
	Section 38(3) and elsewhere makes references to "all business activities". This is ambiguous, in particular given that it is a reference to undertakings that are not	Consideration of the extent to which disclosure or reporting needs to be continuous, or periodic.
	'businesses' but rather the core regulatory and administrative and service provision responsibilities of territorial authorities. Any reporting requirements for the parts	Amend s38(3) and all other occurrences to replace "all business activities" with "other activities".
	of a Council's operation that is not waters it would be better to talk about "activities".	Consider further amendment to s38(3) and other sections in relation to whether it is appropriate or helpful to describe the activities of
	Equally, it is probably both incorrect and unhelpful to describe the core regulatory, administrative and other functions of territorial authorities as the "supply of	territorial authorities as the "supply of goods or services".
	goods or services". They are certainly not goods and services of the nature that are covered by the Consumer Guarantees Act or other fundamentals of sale and	
	purchase arrangements.	

Further Information and Hearings

- **53.** Should Parliament's Finance and Expenditure Committee require clarification of the submission from Hamilton City Council, or additional information, please contact **Andrew Parsons** (General Manager Infrastructure and Assets) on **07 838 6896** or **021 791 612**, or email andrew.parsons@hcc.govt.nz in the first instance.
- **54.** Hamilton City Council representatives **do wish to speak at the hearings** for the Local Government (Water Services Preliminary Arrangements) Bill.
- **55.** We also welcome the opportunity, if available, to have further discussions around the key areas of this submission with Parliament's Finance and Expenditure Select Committee.

Yours faithfully

Lance Vervoort
CHIEF EXECUTIVE

Hamilton City Council Garden Place, Private Bag 3010, Hamilton

/HamiltonCityCouncil

@hamiltoncitycouncil

07 838 6699

hamilton.govt.nz